

Review article

Contents lists available at ScienceDirect

Computer Science Review

journal homepage: www.elsevier.com/locate/cosrev



Systematic literature review: Quantum machine learning and its applications



David Peral-García^{a,*}, Juan Cruz-Benito^b, Francisco José García-Peñalvo^c

a Expert Systems and Applications Laboratory - ESALAB, Faculty of Science, University of Salamanca, Plaza de los Caídos s/n, 37008 Salamanca, Spain

^b IBM Quantum, IBM T.J. Watson Research Center, Yorktown Heights, NY 10598, USA

GRIAL Research Group, Department of Computers and Automatics, Research Institute for Educational Sciences, University of Salamanca, Paseo de Canalejas,

169. Salamanca 37008. Spain

ARTICLE INFO

Keywords: Quantum machine learning Quantum computing Systematic literature review

ABSTRACT

Ouantum physics has changed the way we understand our environment, and one of its branches, quantum mechanics, has demonstrated accurate and consistent theoretical results. Quantum computing is the process of performing calculations using quantum mechanics. This field studies the quantum behavior of certain subatomic particles (photons, electrons, etc.) for subsequent use in performing calculations, as well as for large-scale information processing. These advantages are achieved through the use of quantum features, such as entanglement or superposition. These capabilities can give quantum computers an advantage in terms of computational time and cost over classical computers. Nowadays, scientific challenges are impossible to perform by classical computation due to computational complexity (more bytes than atoms in the observable universe) or the time it would take (thousands of years), and quantum computation is the only known answer. However, current quantum devices do not have yet the necessary qubits and are not fault-tolerant enough to achieve these goals. Nonetheless, there are other fields like machine learning, finance, or chemistry where quantum computation could be useful with current quantum devices. This manuscript aims to present a review of the literature published between 2017 and 2023 to identify, analyze, and classify the different types of algorithms used in quantum machine learning and their applications. The methodology follows the guidelines related to Systematic Literature Review methods, such as the one proposed by Kitchenham and other authors in the software engineering field. Consequently, this study identified 94 articles that used quantum machine learning techniques and algorithms and shows their implementation using computational quantum circuits or ansatzs. The main types of found algorithms are quantum implementations of classical machine learning algorithms, such as support vector machines or the k-nearest neighbor model, and classical deep learning algorithms, like quantum neural networks. One of the most relevant applications in the machine learning field is image classification. Many articles, especially within the classification, try to solve problems currently answered by classical machine learning but using quantum devices and algorithms. Even though results are promising, quantum machine learning is far from achieving its full potential. An improvement in quantum hardware is required for this potential to be achieved since the existing quantum computers lack enough quality, speed, and scale to allow quantum computing to achieve its full potential.

1. Introduction

Currently, there is an ongoing challenge in the scientific world to create quantum computers capable of substantial advantages for certain problems compared to conventional computers. The automated tools improvement and methods to assist in the simulation and design of the corresponding applications are required along with the device's development. Otherwise, one could end up in a situation where one has powerful quantum computers but very few adequate means to use them [1]. On the other hand, scientific progress in materials creation,

hardware fabrication, and disciplines such as error correction and compilation have enabled us to create large-scale and increasingly fault-tolerant quantum computers [2]. Some of the main objectives to be achieved are improving simulation times in chemical compounds, leading to substantial reductions in drug creation times. Also, the creation of complex cryptographic systems will allow the creation of secure computers, guaranteeing internet security for all users and the design of new artificial intelligence algorithms currently housed in most devices, from better prediction and recommendation systems to new industrial ones support systems.

Corresponding author. E-mail addresses: daveral@usal.es (D. Peral-García), juan.cruz.benito@ibm.com (J. Cruz-Benito), fgarcia@usal.es (F.J. García-Peñalvo).

https://doi.org/10.1016/j.cosrev.2024.100619

Received 8 January 2022; Received in revised form 6 December 2023; Accepted 18 January 2024

1574-0137/© 2024 The Author(s). Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).



Fig. 1. Percentage of articles by source.

Since the 90s, we have observed the proposal and development of various highly relevant algorithms that demonstrate the computational advantage over classical computation. Some of them are Grover's search algorithm [3] and Shor's algorithm [4] for integer factorization in polynomial time, both of which significantly outperform conventional machines. Recently, the application area of quantum algorithms has expanded dramatically and provided efficient methods in areas, such as chemistry [5–7], communications [8], solving systems of linear equations [9], physics simulations [10,11], cybersecurity [12–14] and machine learning [15,16]. One of the most emerging areas has been the machine learning field. With existing quantum devices and algorithms, quantum algorithms have already improved some classical processes, and, in contrast, classical machine learning is used to enhance quantum procedures.

Within the application of classical machine learning techniques for improvement of the quantum world, recent studies show the detection of quantum entanglement with unsupervised training in fully and partially entangled structures [17]. Furthermore, deep learning techniques allow us to reduce the noise produced in quantum systems [18] or determine the structural properties and molecular dynamics in the quantum scenario [19]. Regarding the design and implementation of hybrid deep learning algorithms, the use of hybrid graphical convolutional networks (OGCNN) has been proposed, which, compared to quantum convolutional neural networks, the classical multilayer perceptron (MLP) and classical convolutional networks, can achieve better performances, for example, in high-energy physics (HEP) [17]. In the finance application field, hybrid neural networks have been used for predicting Gross Domestic Product growth [20]. In the healthcare field, the use of Boltzmann machines for classifying patients with lung cancer [21].

Bringing all these ideas together, the topic of this article is the link between the different types of quantum machine learning algorithms and their applications: What types of algorithms are used? Where are these algorithms implemented? We will address these questions based on the literature published in these research fields. To achieve these objectives, we conducted a systematic review of the literature. This paper is organized as follows: Section 2 presents the systematic review protocol, including its different aspects and necessary steps. Section 3 presents the systematic review results, Section 4 discusses the topic, and Section 5 presents the conclusions.

2. Materials and methods

To identify, evaluate, and interpret the studies available in the literature and taking into account the research questions proposed by the authors, the methodology proposed by Kitchenham et al. [22] is used, thus organizing it into planning, conducting, and reporting the study. The structure of this section will be distributed according to [23–25]. First, the databases will be selected, then the research questions will be defined as well as the inclusion and exclusion criteria and the definition of the search string, and finally, the review process will be carried out.

While working on this systematic literature review, we used Parsifal (https://parsif.al/) to manage traceability and Mendeley (https://www. mendeley.com/) as the bibliographic manager. On the other hand, two bibliographic reviews have been taken as a reference to structure this study, one of them from a general perspective of quantum computing [26] and another one from a more specific context of quantum machine learning [2]. Furthermore, to facilitate the reproducibility of this study, we provide a public record of the review procedure and steps in https: //github.com/DaveHub5/SLR_Quantum_Machine_Learning. There, the readers can observe the reviewed articles' status through the PRISMA diagram's different phases [27].

We collected in total 5497 articles [Fig. 1], 1962 obtained from IEEE Digital Library, 1261 from Web of Science, 2224 from Science Direct, 45 from IBM Quantum Network Papers, and 5 articles from other sources published between 2017 and 2022. IBM Quantum Network Papers is an index maintained by IBM https://airtable.com/shr5QnbLgraHRPx35/tblqDKDgMVdH6YGSE. The query strings used for the search process were "quantum machine learning", "quantum deep learning", or "quantum neural network*". At the end of the study, we identified 94 articles that met the inclusion criteria, as we explain below. A summary of the identification and selection of the articles included in this systematic literature review can be seen in the PRISMA diagram [Fig. 2].

2.1. Review planning

This review attempts to show solutions and improvements for the development of quantum circuit simulation tools, quantum circuit design, and the creation and adaptation of different algorithms for quantum computation. The paper aims to serve as an introduction to this



Fig. 2. PRISMA diagram.

emerging field, along with a discussion of recent advances and presenting the problems that remain to be solved [26].

The field of quantum computing is a recent field compared with classical computing. For this reason, we will focus on the advances and applications of the last six years (2017-2023). We will compare the different techniques used as well as the challenges. The journals used for the articles retrieval have been IEEE Digital Library, Web of Science, and Science Direct. Also, an index of articles related to the cited topic has been included. These databases were selected according to the following requirements: the database is general-purpose journals with a large number of articles, the database admits full-length searches or searches only in specific fields of the works, is one of the most relevant in the research area of computer science, and the database is available for us, the licenses held by the University of Salamanca. IBM Quantum Network Papers is open access to everyone but is notably different, it is an article collection that includes particular articles of interest to the company or includes work that uses IBM Quantum hardware and tools. We include this not-indexed database with ensured contrasted quality content because essential articles are not included in the other databases.

2.2. Research questions

This article will focus on and try to answer the following questions.

- RQ1: What type of algorithm is used?
- RQ2: Do they use hybrid or pure quantum algorithms?
- RQ3: What are the application domains?
- RQ4: What are the metrics used to measure the algorithm's success rate or solution proposed? What are the metric values reported?
- RQ5: Are the experiments carried out on simulators or real quantum devices?

After the research questions are defined, we used the PICOC method proposed by [28] to define our review scope:

- Population (P): The target group for the investigation. In this study: Quantum Machine Learning.
- Intervention (I): specifies the investigation aspects or issues of interest for the researchers. In our case, those aspects or issues provide support or analyze QML processes.
- Comparison (C): the aspect of the investigation to which the intervention is being compared. No comparison intervention has been planned in our study.
- Outcomes (O): the effect of the intervention. We are looking for Quantum Machine Learning improvements and implementations.

Relation between paper and used algorithm.

Algorithm	[29]	[<mark>30</mark>]	[31]	[<mark>32</mark>]	[33]	[<mark>34</mark>]	[35]	[<mark>36</mark>]	[37]	[38]	[<mark>39</mark>]	[40]	[41]	[42]	[43]	[<mark>34</mark>]	[44]	[45]	[46]	[47]	[48]	[<mark>49</mark>]	[50]	[51]	[52]
vVQC	Х			Х																					Х
HQA		Х																							
QRC			Х																						
QMCC				Х																					
QSVM-Kernel					х																				
RQNN						Х	Х	Х																	
HKNN									Х																
OrthogonalNN										Х	Х	Х													
QFS-Net													Х												
CMN														Х											
QDCNN															Х										
QBP																х									
ffNN							х										х								
LSTM-QNN																		Х	Х				Х	Х	
QGAN																				Х	Х	Х			

• Context(C): the setting or environment of the investigation. In our case, they are those environments related to QML (in the industry, universities, etc.).

2.3. Inclusion and exclusion criteria

Once these questions have been defined, the following inclusion and exclusion criteria are defined.

Inclusion criteria:

- IC1: The solutions provided by the articles are either applied in software or can be simulated or implemented in simulators or quantum devices AND
- · IC2: The papers are in a Journal, Book, Preprint or Conference AND
- · IC3: The papers are written in English or Spanish AND
- · IC4: The papers provide a practical solution

Exclusion criteria:

- · EC1: The solutions provided by the articles are not applied in software and cannot be simulated or implemented in simulators or quantum devices OR
- · EC2: The papers are not in a Journal, Book, Preprint or Conference OR
- EC3: The papers are not written in English or Spanish OR
- · EC4: The papers do not provide a practical solution

2.4. Query string

The first step is to collect all the articles from 2017 to 2023, we have selected this period because it corresponds to the beginning of the rise and growth of quantum devices, as well as the proliferation and possible use of noisy quantum computers or noisy intermediate-scale quantum (NISQ) devices [53]. To carry out this process, we define some keywords as follows: quantum deep learning, quantum machine learning, quantum neural network.

For this purpose, we use the following queries:

- IEEE Digital Library: (All Metadata : quantum machine learning) OR (All Metadata: quantum deep learning) OR (All Metadata : quantum neural network*) Filters Applied: 2017-2023
- · Web of Science: (quantum machine learning OR quantum deep learning OR quantum neural network*) Time period: 2017-2022
- · Science Direct: quantum machine learning or quantum deep learning or quantum neural network From 2017 to 2022

We observe no possible query in the case of IBM Quantum Network Papers. However, we applied a filter in the field Research Domain of the type Machine Learning because the index of articles is specific to the topic of quantum computing, so there is no need to indicate the word quantum itself.

2.5. Review process

When exporting the information from the articles, we encountered a problem in the case of IBM Quantum Network Papers, which does not allow export to BibTeX format, although it has a link to the arvix.org page.

From these queries, we obtained 5497 articles, 1962 from IEEE Digital Library, 1261 from Web of Science, 2224 from Science Direct, 45 from IBM Quantum Network Papers, and 5 articles from other sources. After removing duplicates, we got 1583 unique records. We searched through the titles for those referring to the following words quantum machine learning, quantum deep learning, quantum neural network, leaving us with 382. With this manageable number, we proceeded to read the abstracts, obtaining 338 articles. With a more exhaustive review, we reduce the list to 163, to which we will apply quality criteria. These criteria are as follows:

- · QC1. Are the research aims specified?
- · QC2. Is the solution implemented?
- · QC3. Does the study provide accurate results?
- QC4. Does the study show the applied quantum circuit used?

We apply these quality criteria for every paper. Each article should have a score \geq 3.0 out of 4.0 points to be selected, considering one point for each item in the quality criteria list. We obtained a set of 70 articles as the final result.

3. Results

3.1. Algorithm types

Regarding RQ1, Table 1 show the distribution of algorithms and the associated papers. In order to answer RQ2, in this section, we will focus on the different types of Quantum Machine Learning algorithms. We will give a brief explanation of their mathematical basis and implementation of each one of them. The first step of the following algorithms is data encoding, i.e., transforming bits into qubits, a type of information understandable by quantum computers. The second is the creation of the quantum circuit or ansatz, in which the different quantum logic gates are applied to the states. Finally, the reading and measurement of the qubits (see Table 2).

Table	2
-------	---

Relation between	elation between paper and used algorithm.														
Algorithm	[54]	[55]	[56]	[57]	[58]	[59]	[60]	[61]	[62]	[63]	[64]	[65]	[66]	[67]	
RQNN	Х	Х													
QREDNN			х												
QCNN				Х	Х									Х	
VQA						Х									
QPSO							Х								
Transformers								Х							
QSVM									Х						
VQE										Х					
QBM											Х	Х			
QMonteCarlo													Х		

3.1.1. Quantum Boltzmann machines (QBM)

A QBM is defined by a Hamiltonian model of the form $H_{\theta} = \sum_{i=0}^{p-1} \theta_i h_i$ where $\theta \in \mathbb{R}^p$ y $h_i = \bigotimes_{j=0}^{n-1} \sigma_{j,i}$ for $\sigma_{j,i} \in \{I, X, Y, Z\}$ acting in the *j*th qubit. Equivalent to classical Boltzmann machines [68], QBMs are typically represented by an Ising model.

The nodes of the network, given by the Pauli terms $\sigma_{j,i}$, are defined concerning certain subsets of qubits. The qubits that determine the output of the QBM are the visible qubits, while the others correspond to the hidden qubits. The probability of measuring a configuration v of the visible qubits is defined with respect to the $\Lambda_v = |v\rangle \langle v| \otimes I$ in the quantum Gibbs state [64,65].

$$\rho^{Gibbs} = \frac{e^{-H_{\sigma}/(k_B T)}}{Z} \tag{1}$$

with $Z = Tr[e^{-H_{\sigma}/(k_BT)}]$, i.e., the probability of measuring $|v\rangle$ is given by

$$p_{v}^{QBM} = Tr[\Lambda_{v}\rho^{Gibbs}]$$
⁽²⁾

The Gibbs state ρ^{Gibbs} describes the probability of a system being in a certain state as a function of that state's energy and the system's temperature *T*.

3.1.2. Variable depth quantum circuits (vVQC)

In [29], the authors aim to automatically change the quantum circuit structure according to the score of the qBAS metric [69], a complementary metric designed for benchmarking quantum–classical hybrid systems, to solve the problem of the inflexibility of the Variational Quantum Classifier (VQC) model [70]. Furthermore, in this paper, the depth of the quantum circuit (the number of quantum circuit layers) is considered a parameter during training.

The quantum circuit block is taken $U_l(\theta_l)$ as a layer, which is constructed using the circuit block parameterized in [Fig. 3]. It is constructed using $e^{-i\theta\sigma}$ rotation gates, where σ is chosen from the set of $\{X, Y, Z\}$ Pauli gates, which are applied at each qubit. The θ degree of the rotation is the circuit parameter to be tuned.

Also, in [52] a framework [Fig. 4] to train asynchronously these circuits is presented. The quantum asynchronous advantage actor-critic (QA3C) framework includes a global shared parameters and multiple parallel workers. The results of the gradients of each local agent are stored in the process-specific memories; after that, they are uploaded to the global shared memory, where the global model parameters are updated. To conclude, the updated global model parameters are then immediately downloaded to the local agent that just uploaded its own gradients.

3.1.3. Hybrid quantum autoencoders (HQA)

A variation of the Quantum Amplitude Estimation (QAE) algorithm [71] called hybrid quantum autoencoder (HQA) is proposed in [30]. This model incorporates both classical machine learning, in the form of neural networks (ANNs), and quantum machine learning, using quantum neural networks (QNNs) based on parameterized quantum circuits (PQC).

The general design of the model, which is a combination of an encoder that takes a quantum state from Hilbert space $H_2^{\otimes n}$ to a subset



Fig. 3. VQC framework [29].



Fig. 4. Quantum asynchronous advantage actor-critic (A3C) learner [52].

of real vector space V of dimension v = dim(V), and a decoder that performs the inverse of that operation.

Although the functional form of the encoder and decoder are defined, the models themselves are not specified. As seen in [Fig. 5], the ϵ encoder is a quantum circuit parameterized by a vector α . The circuit receives some $|\psi_{in}\rangle$ state and applies the unitary $U_1(\alpha)$ on the combined system of the input state with (v - n) auxiliary qubits.

3.1.4. Quantum reservoir computing (QRC)

Quantum reservoir computing (QRC) uses a dynamic system to execute temporal information processing tasks [31]. The main objective of this task, called forecasting, is creating a function to transform an input sequence (time series) into a target output sequence for time-series prediction and pattern classification.



Fig. 5. Hybrid Quantum Autoencoder (HQA) [30].



Fig. 6. Quantum variational circuit model [32].

In the QRC framework, the dynamical system is given by

$$\rho_t = T_u (\rho_t - 1), \tag{3}$$

where ρ_t is the density operator representing a state at time t, and T_{u_t} is an input-dependent, fully positive, trace-preserving map (CPTP) describing the time evolution of the QR system. The considered CPTP map is given by

$$\rho_t = e^{-iH\tau} (\rho_{input} \otimes Tr_{input}(\rho_{t-1})) e^{iH\tau}, \tag{4}$$

where $\rho_{input} = |\psi_{u_t}\rangle\langle\psi_{u_t}|with|\psi_{u_t}\rangle = \sqrt{1-u_t}|0\rangle\langle0| + \sqrt{u_t}|1\rangle\langle1|$. An auxiliary qubit takes the input value $u_t \in [0, 1]$, and then the whole QR system is transformed in time by the input-independent unitary operator $e^{-iH\tau}$.

3.1.5. Quantum multiclass classifier (QMCC)

In [32], a variational circuit designed to classify the dataset with four features is proposed [Fig. 6]. The operations performed in the variational circuit are as follows. The four qubits of the circuit are initialized in the $|0\rangle$ state. Next, the Hadamard gate is applied individually to place the qubits in a superposition of $|0\rangle$ and $|1\rangle$. Secondly, a unitary operation is applied to each qubit with a unitary square matrix designed for state preparation. In this way, the classical data (bits) are encoded into qubits.

After state preparation, the variational circuit is designed using multiple layers of interleaved rotational gates in data qubits and auxiliary qubits with adjustable parameters using an optimization technique. The architecture of the variational circuit model and the original implementation of the circuit, which consists of seven layers, is depicted in [Fig. 7]. Finally, auxiliary qubits are measured, and resulting qubits are processed to obtain the class label.

3.1.6. Support vector machines with a quantum kernel estimator (QSVM-kernel)

The quantum support vector machine QSVM and the quantum kernel estimator (QSVM-Kernel) [33] exploits the quantum state space



Fig. 7. Architecture of the quantum variational circuit [32].

as a feature space to compute kernel inputs efficiently. This algorithm maps the classical data \vec{x} nonlinearly to a quantum state of N qubits by applying a quantum circuit $\Gamma_{\Phi(\vec{x})}$ to the initial state $|0^{\otimes N}\rangle$:

$$|\boldsymbol{\Phi}(\vec{x})\rangle = \Gamma_{\boldsymbol{\Phi}(\vec{x})} |0^{\otimes N}\rangle \tag{5}$$

The quantum circuit $\Gamma_{\Phi(\bar{x})}$ gives rise to a 2^N dimensional feature space (N is the number of qubits), which is hard to estimate classically. This circuit consists of two repeated layers.

$$\Gamma_{\boldsymbol{\phi}(\vec{x})} = U_{\boldsymbol{\phi}(\vec{x})} H^{\otimes N} U_{\boldsymbol{\phi}(\vec{x})} H^{\otimes N}$$
(6)

where H is a Hadamard gate and $U_{Phi(\bar{x})}$ is a unitary operator that encodes the classical input data [Fig. 8].

The kernel entries are evaluated for the training data and used to find a separation hyperplane in the training phase. Then, in the test phase, the kernel inputs are evaluated with new data \vec{x} and with the training data, which are used to classify the new data \vec{x} according to the separation hyperplane.



Fig. 8. Quantum circuit of support vector machine [33].



Fig. 9. Quantum neuron model [34].

The kernel inputs are evaluated on quantum computers, while the optimization of the separation hyperplane and the classification of the data are performed on classical computers, as in a classical SVM.

Regarding quantum kernels, in [72], the authors introduced projected quantum kernels and outperformed all tested classical models in prediction error. First, they use principal component analysis [73] to transform each image into a *n*-dimensional vector. Second, they embed this vector into a quantum circuit using three techniques, a separable rotation circuit [74–76], an IQP circuit [77], and a Hamiltonian evolution circuit.

3.1.7. Quantum neural networks QNN

Quantum neural networks are composed by quantum neurons, quantum neuron model is described in [34]. The input of this model is the quantum state $|x_i\rangle$, and the output is denoted as real number *o* [Fig. 9]. The value of the quantum weight $R(\theta_i)$ is the quantum rotation gate $R(\theta) = \begin{bmatrix} \cos\theta & -\sin\theta \\ \sin\theta & \cos\theta \end{bmatrix}$, Controlled-NOT gate $U(\tau) = C(f(\tau))$, where *f* is the sigmoid function, and the function C(x) is $C(k) |\phi\rangle = \begin{bmatrix} \cos(\frac{\pi}{2}m - \alpha) \\ \sin(\frac{\pi}{2}m - \alpha) \end{bmatrix}$ [Fig. 9].

Once the quantum neuron model is introduced, the model of quantum neural networks is presented, [35,36] where classical input and weight vectors of size *m* can be encoded in quantum hardware using only $N = log_2m$ qubits. Relative quantum phases (i.e., ±1) in superpositions of base states can be used to encode collections of classical data.

The network training can be done in two ways, either by a global variational training, where the quantum artificial neuron is obtained by inverting a non-trivial quantum state $|\psi_w\rangle$ back to the base state $|1\rangle^{\otimes N}$, or by local variational training, in which global transformation $V(\vec{\theta})$ is divided into successive layers $V_j(\vec{\theta}_j)$ of decreasing complexity and size. Each layer is trained separately, according to a cost function applied to obtain the desired final state.

Transfer learning for QNN is proposed in [78] and in [67]. The authors use an effective strategy for small QCNNs in the noisy intermediatescale quantum era. Allowing QCNN to solve complex classification problems by utilizing a pre-trained classical convolutional neural network.



Fig. 10. Test swap circuit [37].



Fig. 11. QKNN circuit [37].

3.1.8. Hybrid k-neighbors-nearby model (HKNN)

In quantum states, the overlap between two states acts as a similarity measure analogous to the Euclidean distance, and this overlap is found through a simple circuit known as a swap-test, as shown in [Fig. 10]. This circuit can be used to evaluate the distances between classical vectors in KNN algorithms [37].

This circuit takes input data that is in the state $|0\rangle$ or $|1\rangle$, of the form $|q_1\rangle |q_2\rangle \dots |q_n\rangle$. Then, successive SWAP-controlled gates or Fredkin gates controlled with an auxiliary qubit are applied. The reference state is sent to unitary qubit *V* [Fig. 11], generating a test circuit instead of the reference circuit.

Comparing the qubit of the reference state to its corresponding one sent to the test circuit, if the output state of the test circuit $|\psi\rangle$ is the same as the reference state $|\phi\rangle$, then the probability of measuring the auxiliary bit to be in the state $|0\rangle$ is one, however, if the states are slightly different, the probability of measuring the auxiliary bit in the zero state is determined by :

$$P(|0\rangle_{anc}) = \frac{1}{2} + \frac{1}{2} |\langle \psi | \phi \rangle|^2,$$
(7)

The generalization for multiple qubits is given by:

$$P(|0\rangle_{anc}) = \frac{1}{2^n} + \frac{1}{2^n} \sum_{i=1}^n |\langle \psi_i | \phi_i \rangle|^2,$$
(8)

where the respective *i*th qubits of the two states are summed, and *n* is the number of qubits introduced into the test circuit. This probability is the overlap between the two states. Using *d* different pairs of input and reference states, a classical *d*-dimensional vector \mathbf{S} is constructed containing the measured probability for each set of states such that:

$$\mathbf{S} = [P_1, P_2, \dots, P_d] \tag{9}$$

3.1.9. Orthogonal neural networks

Orthogonal neural networks have recently been introduced as a new neural network type that enforces orthogonality in weight matrices [38, 39]. Several classical gradient descent methods have been proposed to preserve orthogonality while updating weight matrices, but these techniques suffer from long runtimes and provide only approximate orthogonality.



Fig. 12. Quantum mapping in two qubits [40].

 $-H \bullet R_y(+\theta/2) \bullet H - H \bullet R_y(-\theta/2) \bullet H -$

Fig. 13. Decomposition of the RBS door [40].

The idea of orthogonal neural networks (OrthoNNs) is to add a

constraint to the weight matrices corresponding to the layers of a neural

network, with algorithms such as the Stiefel manifold [38]. However,

Then, map that result to the amplitude in the kth ground state

$$|a_k\rangle = \left(\frac{2\pi}{D}f_{QNNM}(x_k)\right) \tag{13}$$

An example of such a network is the QFS-Net architecture [41], where the information processing unit is the qutrit. The qutrit neurons in each layer are obtained using a transformation gate T, and the interconnection weights are mapped onto the qutrits using Hadamard gates (H). The rotation angle (pink arrow in [Fig. 14]) is adjusted by obtaining the relative difference of the information between each candidate qutrit neuron and the neighboring qutrit neuron of the same layer used in the rotation gate to update the interconnections between layers. A new sigmoid activation function guides the self-propagation and counter-propagation mechanisms of the QFS network.

3.1.11. CNOT neural networks (CMN)

Although quantum neural network models exist, their computational complexity may require specific unitary transformations to simulate the neuron's activation function or a large number of auxiliary qubits to perform the desired transformations. To solve some of these problems, a quantum neural network model called the CNOT measured network (CMN) is presented in [42].

The CMN uses only CNOT quantum gates and the measurement operator, which are very simple to implement in any quantum computing technology. The CMN can, using only these two simple operators, accommodate universal operators such as AND and OR while maintaining an optimized learning rate and a constant number of auxiliary qubits.

The main objective of the proposed model allows reducing the computational complexity required and minimize the number of different computational elements. In order to achieve this objective, only discrete Boolean functions are designed, and models with a larger number of layers with simple transfer functions are chosen.

The phases of the network creation process would be the initialization procedure for an arbitrary quantum state, the implementation of the CNOT gate, and the addition of the measurement operator.

In the information processing stage, two types of qubits are distinguished, data qubits and weight qubits. Data qubits x_1 and x_2 are initialized to a state corresponding to the value of the input variables. For example, $|x_1\rangle = |0\rangle$. The weights are initialized with a random value as $w_i = \frac{1}{\sqrt{3}} |0\rangle + \frac{\sqrt{2}}{\sqrt{3}} |1\rangle$ [Fig. 15].

The data qubits are then interleaved with the weight qubits using the *CNOT* quantum gate as follows:

$$\begin{aligned} |\phi\rangle &= CNOT \ |w_{ji} \otimes x_i\rangle \\ &= \alpha_{w_{ii}} \alpha_{x_i} \ |0\rangle_{w_{ii}} \otimes |0\rangle_{x_i} \pm \beta_{w_{ii}} \beta_{x_i} \ |1\rangle_{w_{ii}} \otimes |1\rangle_{x_i} \end{aligned}$$
(14)

The output function can be expressed using the notation of the density operator as a reduced density operator, described by

$$\rho = \sum_{i} p_{i} |\psi_{i}\rangle \langle\psi_{i}| \tag{15}$$

And the indices *i* the measurement of the qubits $|x_0w_0...x_nw_n\rangle$ can be represented by the trace of the density matrix:

$$\rho^{o} = \sum_{i} p_{i} tr(|\psi_{i}\rangle \langle \psi_{i}|) \otimes \sum_{i} p_{j} |o_{j}\rangle \langle o_{j}|$$

$$\rho^{o} = \sum_{i} p_{j} |o_{j}\rangle \langle o_{j}|$$
(16)

the main difficulty with OrthoNNs is preserving the orthogonality of the matrices while updating during gradient descent.

A new type of neural network layer called Pyramidal Circuit is introduced in [40], which implements orthogonal matrix multiplication. It allows gradient descent with perfect orthogonality with the same execution time as a standard layer.

The proposed quantum circuit implements a fully connected neural network layer with an orthogonal weight matrix, using only one type of quantum gate, the Reconfigurable Beam Splitter (RBS) gate. This two-qubit gate is parameterizable with an angle $\theta \in [0, 2\pi]$. Its matrix representation is given by:

$$RBS(\theta) = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & \cos\theta & \sin\theta & 0 \\ 0 & -\sin\theta & \cos\theta & 0 \\ 0 & 0 & 0 & 1 \end{pmatrix}$$
(10)
$$RBS(\theta) : \begin{cases} |01\rangle \mapsto \cos\theta |01\rangle - \sin\theta |10\rangle \\ |10\rangle \mapsto \sin\theta |01\rangle + \cos\theta |10\rangle \end{cases}$$

Starting from two qubits, one in the $|0\rangle$ state and the other in the $|1\rangle$ state, the qubits may or may not be swapped in superposition. The $|0\rangle$ qubit remains in its thread with the amplitude *cos* or, conversely, is swapped with the other qubit if the amplitude is $-sin\theta$ while the new thread is above $(|01\rangle \mapsto |10\rangle)$, or $+sin\theta$ if the new thread is below $(|01\rangle \mapsto |10\rangle)$. In the other two cases $(|00\rangle \mapsto |11\rangle)$ the gate *RBS* acts as an identity [Fig. 12].

This gate can be implemented either as a single gate known as the *FSIM*, or using four Hadamard gates, two R_y rotation gates, and two two-bit *CZ* gates [Fig. 13]:

3.1.10. Quantum fully self-supervised neural networks (QFS-net)

In [41], a quantum neural network model based on qudits is used. A qudit is a multilevel quantum system (D > 2) represented by D basis states: $|0\rangle$, $|1\rangle$, $|2\rangle$, ..., $|D - 1\rangle$, in which the superposition of these basis states would be represented by:

$$|\psi\rangle = a_0 |0\rangle + a_1 |1\rangle + a_2 |2\rangle + \dots + a_{D-1} |D-1\rangle = \begin{bmatrix} a_0\theta \\ a_1\theta \\ \dots \\ a_{D-1} \end{bmatrix}$$
(11)

with the normalization criterion $|a_0|^2 + |a_1|^2 + \dots + |a_{D-1}|^2 = 1$.

In this type of qubit-based quantum neural network model, the classical inputs to the network become D-dimensional given by x_k . A sigmoid function is applied to the quantum states $[0, (2\pi/D)]$ as long as the *k*th input is the function $f_{QNNM}(x_k)$, yielding a binary classical result [0, 1].

$$f_{QNNM}(x_k) = \frac{1}{1 + e^{-x_k}}$$
(12)



Fig. 14. QFS-Net [41]. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)



Fig. 15. CNOT quantum neural network [42].

3.1.12. Quantum convolutional deep convolutional neural networks (QD-CNN)

Another type of algorithm is the quantum adaptation of convolutional networks. For example, in [43], we can see an implementation of a deep convolutional quantum convolutional neural network (QDCNN).

The first step of QDCNN is preparing the classical input image in a quantum state. Given a $2^n \times 2^n$ classical gray input image *I*, it can be described as a matrix:

where $N = 2^n$ and the elements $c_{x,y} \in [0, 255]$ represents the value of the corresponding pixel at position $(x, y) \in [0, N-1]^{\otimes 2}$

Analogous to the classical situation, the QDCNN architecture consists of several successive quantum convolutional layers and a quantum classifier layer.

First, all qubits in the quantum register are initialized as $|0\rangle$ state. Afterward, Hadamard gates are applied to the regL quantum register and the QRAM algorithm to the regC and regK quantum registers to prepare the input image and the convolutional kernel, respectively. Quantum multiplication is applied to the regC and regK registers. A bifurcation qubit *qbitF* and a rotation qubit *qbitR* are added [Fig. 16].

$$U_{QRAM} = (\{d_j\}) |j\rangle |0\rangle \mapsto |j\rangle |d_j\rangle \tag{18}$$

The QPE algorithm of the G oracle [43][Fig. 17] is implemented on the *regA* obtaining an estimate of the quantum amplitude (QAE). Finally, an additional quantum register is introduced to make the linear quantum theory compatible with the nonlinear DCNN model, and a nonlinear modulus $U_{f(x)}$ is used.

3.1.13. Quantum backpropagation neural networks (QBP)

Based on the quantum neuron model and the structure of the classical back-propagation (BP) neural network, the QBP feedforward neural network [Fig. 18] is constructed in [34].

In this model, the input and hidden neurons are quantum, and the output neurons are classical. The input vector is $X = [|x_1\rangle |x_1\rangle |x_i\rangle ... |x_n\rangle]^T$, the final output vector of the network is $O = [o_1o_2 ... o_k ... o_l]^T$ and the weight matrix $V = (v_{i,j})_{nm}$, where v_{ij} is the weight between the *i*th input node and the *j*th hidden node.

Since the training samples are always real numbers, a specific method for the transfer of real numbers to quantum states is defined. The vector of real samples is $X = [x_1 x_2 \dots x_i \dots x_n]^T$, thus we can obtain the corresponding input quantum vector $[|x_1\rangle |x_1\rangle |x_1\rangle \dots |x_n\rangle]^T$, where

$$X_{i} = \begin{bmatrix} \cos(2\pi) \cdot sigmoid(x_{i}) \\ \sin(2\pi) \cdot sigmoid(x_{i}) \end{bmatrix}$$
(19)

To avoid the local optimization problem in the gradient descent calculation, a genetic algorithm (GA) is used. In contrast to the GA, the QGA [79] can maintain the diversity of the population due to the quantum superposition state, simplifies the calculation, and reduces the operation steps by using the quantum rotational gate. By combining the QGA algorithm and BP algorithm, the training error of the QBP neural network can be minimized, and we can prevent the training from falling into a local optimization situation.

In this model, each chromosome is encoded by qubits in the following way

$$q_{j}^{t} = \begin{bmatrix} \alpha_{11}^{t} & \dots & \alpha_{1k}^{t} & \alpha_{21}^{t} & \dots & \alpha_{2k}^{t} & \dots & \alpha_{m1}^{t} & \dots & \alpha_{mk}^{t} \\ \beta_{11}^{t} & \dots & \beta_{1k}^{t} & \beta_{21}^{t} & \dots & \beta_{2k}^{t} & \dots & \beta_{m1}^{t} & \dots & \beta_{mk}^{t} \end{bmatrix}$$
(20)

where q_j^t is the *j*th chromosome of the *t*th generation, *k* represents the number of qubits encoded by each gene, m is the number of genes in the chromosome.



Fig. 16. QDCNN-net architecture [43].



Fig. 17. QDCNN-Net G oracle [43].

3.1.14. Feed foward neural networks (ffNN)

The architecture of quantum ffNNs is defined in [35,44]. An ffNN is essentially composed of a set of individual n_i nodes, or artificial neurons, arranged in a sequence of successive L_j layers. Information flows through the network from the input layer to the output layer, traveling through connections between neurons.

Artificial neurons combine the input (\vec{i}) and the weight (\vec{i}) . (\vec{w}) , providing an activation that depends on their scalar product $\vec{i} \cdot \vec{w}$.

Assuming that the quantum register of *N*-qubits is initially in the rest configuration, $|0\rangle^{\otimes N}$, the quantum state of the input vector with a unitary operation is prepared U_i such that $|\psi_i\rangle = U_i |0\rangle^{\otimes N}$. Next, the weight factors of the vector are applied \vec{w} on the input state by implementing another unitary transformation, U_w , subject to the constraint $|1\rangle^{\otimes N} = U_w |\psi_w\rangle$.

When several copies of the quantum register run in parallel, these, and the result of measurements made on them, can be used to feed information about the processing of the input weight to a successive layer.

An abstract representation of the proposed architecture is shown in [Fig. 19].

In [80], another implementation of ffNN is proposed, where the data is introduced in each processing unit joined with the output of the previous processing unit. The data is uploaded multiple times along the circuit by using one-qubit rotations. These rotations are optimized using a classical minimization algorithm.

3.1.15. Long short-term memory neural networks LSTM-QNN

In classical deep learning, Long short-term memory (LSTM) networks are used to detect both short and long-term dependencies in entire sequences of data. One of the implementations of the LSTM cell [Fig. 20] has three inputs; the first one is represented by x_t , where t is the time index of data samples that is also associated with the specific cell of the unrolled chain, the second input h_{t-1} is the 'hidden' state computed at the previous cell (time step) and the third input c_{t-1} is the previous 'cell' state. In addition, the cell has three control state cell gates: an input gate (i), a forget gate (f), and an output gate (o). Also, each gate has the sigmoid σ or the hyperbolic tangent tanh activation function [50].

A hybrid architecture is proposed in [Fig. 21] [51]. It consists of two layers with 32 cells, two fully connected layers, the second one of 10 neurons, to connect with the QNN. The QNN layer is composed using the IQP Ansatz [77] and StronglyEntanglingLayers [70], adding a final output classical layer.

due to the limitation of a 10 qubits layer with 32 cells, it must be followed by a classical layer, which is another dense layer in this example, containing ten neurons, before it can be connected to the QNN with ten qubits. Then, the final predictions may be obtained by adding a fully connected layer after the QNN.

In [45], a metalearning quantum approximate optimization algorithm (MetaQAOA) is proposed for the MaxCut problem [81]. Specifically, a quantum neural network (QNN) in the form of a parameterized quantum circuit and a classical short-term memory (LSTM) neural network is constructed as an optimizer, which can quickly help the QNN [46] to find the approximate optimal parameters of the QAOA.

The main steps of this algorithm are as follows. A uniformly distributed quantum superposition state is prepared as the initial state $|\psi_0\rangle$:

$$|\psi_0\rangle = |+_1\rangle |+_2\rangle \dots |+_n\rangle, \tag{21}$$

where *n* represents the number of nodes in the graph G for the MaxCut problem and $|+\rangle = (1/\sqrt{2})(|0\rangle |1\rangle)$ is an overlapping state.

Next, the Hamiltonian operator \hat{H}_M and the cost Hamiltonian operator \hat{H}_M are selected \hat{H}_{C_1} :

$$\hat{H}_M = \sum_{i \in \eta} \hat{X}_j \tag{22}$$

$$\hat{H}_{C_1} = \sum_{\{j,k\}\in E} \frac{1}{2} (\hat{I} - \hat{Z}_j \hat{Z}_k),$$
(23)

where \hat{H}_M is the Pauli sum \hat{X} in each qubit, \hat{H}_{C1} is diagonal in the computational basis, I, X, Z represent the Pauli operator I, X, Z, respectively and j, k are the two nodes connected by the edge of the graph G.

Thirdly, the unitary operator $\hat{U}(\eta,\gamma)$ applied to this initial state given by

$$\hat{U}(\eta,\gamma) = \prod_{q=1}^{P} e^{-i\eta_q \hat{H}_M} e^{-i\gamma_q \hat{H}_{C_2}}$$
(24)



Fig. 18. Quantum backpropagation neural network [34].



Fig. 19. Feed Foward Neural Network (ffNN) [44].



Fig. 20. LSTM cell [50].

where *P* is the number of times the unitary transformation is executed and the parameters $\{\eta, \gamma\}$ defined by:

 $\vec{\eta} = (\eta_1, \eta_2, \dots, \eta_p)^T,$ $\vec{\gamma} = (\gamma_1, \gamma_2, \dots, \gamma_p)^T.$ (25)

The resulting parameterized circuit is shown at [Fig. 22].

3.1.16. Quantum generative adversarial networks (QGAN)

In these papers [47–49], a quantum version of the Generative Adversarial Network (GAN) [82] is proposed. The input data to the discriminator are the sampled data pairs from the classical training

set, where *y* represents the conditional variable. The generator obtains the conditional variables and the probability distribution of various samples in the training set through $|y\rangle$. Subsequently, the classical discriminator is trained with the training set data and the samples from the quantum generator, whose output feeds back to the quantum generator [Fig. 23].

At the beginning of the training, all parameters of the quantum circuit and the binary classification neural network are given random initial values. The generator and discriminator parameters are alternately optimized during the training process. The discriminator simultaneously judges the training data and the data sampled from the quantum generator.

The generator's ability to match the actual distribution is improved by repeating the circuit in the quantum computing device. Optimization of the generator and discriminator parameters must be performed iteratively until the generator can reconstruct the state distribution of the training set.

$$|y\rangle = \sum_{j=1}^{m} \frac{1}{\alpha_j} |y_j\rangle, \qquad (26)$$

Regarding the quantum generator circuit design, a PQC is generated by implementing single-qubit gates that are used to perform the rotation of the qubits, as well as multi-qubit gates mainly perform the entanglement between the qubits.

The gates R_x and R_z being defined as:

$$R_{x}(\theta) = \begin{bmatrix} \cos(\theta/2) & -i\sin(\theta/2) \\ -i\sin(\theta/2) & \cos(\theta/2) \end{bmatrix}, R_{z}(\theta) = \begin{bmatrix} e^{-i\theta/2} & 0 \\ 0 & e^{i\theta/2} \end{bmatrix}$$
(27)

The XX in [Fig. 24] represents an operation involving two qubits, where one is the control qubit, and the other is the target qubit.

The N_c qubits are only responsible for passing the conditional information to the other N_d qubits and continuing to pass the conditional information to the discriminator in post-processing.

3.1.17. Recurrent quantum neural networks (RQNN)

A comparison of the models of a QNN_{QG} and $RQNN_{QG}$ for an (r-1)th and *r*th measurement round [Fig. 25] is performed in [54]. The input values are represented by $|\psi_{r-1}\rangle|1\rangle$ and $|\psi_r\rangle|1\rangle$, while the output values are denoted by $|Y_{r-1}\rangle$ and $|Y_r\rangle$. The output of the measurement operator *M* in the (r-1)th and *r*th measurement rounds is denoted by $|Y_{r-1}^{(r)}\rangle$ and $|Y_{r}^{(r)}\rangle$.

 $|Y_{n+1}^{(r-1)}\rangle$ and $|Y_{n+1}^{(r)}\rangle$. In [Fig. 25(a)], the structure of a QNN_{QG} for an (r-1)th and *r*th measurement round is depicted, while in [Fig. 25(b)], the structure of an $RQNN_{QG}$ is illustrated.

The advantage of this type of network is that information about gate parameters, and measurement results are transmitted from the (r-1)th measurement round (represented by the dashed gray arrows) to the *r*th measurement round.

Another implementation of RQNN is presented in [55], where the authors proved that, even if the transformations in quantum circuits



Fig. 22. LSTM-QNN parameterized circuit [45].



Fig. 23. Conditional adversarial neural networks [48].



Fig. 24. QGAN quantum circuit generator [48].



Fig. 25. QNN y RQNN [54].



Fig. 26. Quantum recurrent encoding-decoding neural network [56].

– with gates and measurements – will necessarily be linear operations, nonlinear behaviors may occur during these transformations. For example, in

$$\mathbf{R}(\theta) = \exp\left(i\theta \begin{pmatrix} 0 & -i\\ i & 0 \end{pmatrix}\right) = \begin{pmatrix} \cos\theta & \sin\theta\\ -\sin\theta & \cos\theta \end{pmatrix},$$
(28)

we can see that the rotation transformation is a linear operator; meanwhile, the amplitudes of the state, $\cos \theta$ and $\sin \theta$ depend non-linearly on the angle θ .

3.1.18. Quantum recurrent encoding-decoding neural networks (QREDNN)

In the quantum recurrent encoder-decoder neural network (QREDNNN) [56], the attention mechanism is used to reconstruct the encoder and decoder of the QREDNNN simultaneously. With this mechanism, the QREDNNN can suppress the interference of redundant information, obtaining better nonlinear approximation capability.

On the other hand, the quantum neuron is used to construct a new quantum recurrent unit (QGRU), which is proposed in the same paper. The activation values and weights are represented by quantum rotation matrices. The QGRU has a large number of multiple attractors, so it can replace the traditional recurrent unit of the encoder and decoder and improve the generalization ability and response speed of QREDNN.

In addition, the Levenberg–Marquardt (LM) algorithm [83] is introduced to improve the update rates of the rotation angles of quantum rotation matrices and the attention parameters of QREDNNN [Fig. 26].

3.1.19. Quantum vision transformers

In [61], the authors designed the architecture for the Quantum Vision Transformers. First, the network decomposes the image into

patches and pre-processes the set of patches to map each one into a vector. After that, the feature extraction is performed using a transformer layer, and after that, a normalization layer is applied, and the attention mechanism is. When this is done, another normalization layer is applied, followed by Multi-Layer Perceptron (MLP) [Fig. 27].

To conclude, another important aspect in [61] is the usage of the *amplitude encoding* technique, which uses the classical scalar component of the data as amplitudes of a quantum state made of d qubits.

3.1.20. Quantum Monte Carlo

The authors in [66] were the first to apply the quantum Monte Carlo algorithm to problems in economics; they began with a Gaussian sampling and, after that, they made stress testing of banks and solved DSGE models with deep learning.

They define two unitary operators to develop a quantum alternative for the Monte Carlo algorithm [90], *A* for modeling function $p(\cdot)$ and unitary *R* for modeling function $f(\cdot)$ in the classical equation given by

$$\mu := E[f(x)] = \int_{\mathbb{R}^d} p(x)f(x)dx.$$
 (29)

First, defines A considering a register of *m* qubits and the initial state $|0\rangle^{\bigotimes m}$

$$A |0\rangle^{\bigotimes m} = \sum_{i \in x} \sqrt{p(i)} |i\rangle$$
(30)

After that, the R operator plus an additional ancilla qubit is defined

$$R|i\rangle|0\rangle = |i\rangle \left(\sqrt{1 - f(i)}|0\rangle + \sqrt{f(i)}|1\rangle\right)$$
(31)

And finally, the output state as

$$|\chi\rangle := F |0\rangle^{\bigotimes m+1} = \sum_{i \in X} \sqrt{p(i)} |i\rangle \bigotimes \left(\sqrt{1 - f(i)} |0\rangle + \sqrt{f(i)} |1\rangle\right)$$
(32)

3.2. Application domains and metrics

Regarding RQ3, the different applications of the models explained in the previous section are presented, distinguishing between applications with a classical purpose, i.e., those that help improve classical processes, and applications with a quantum purpose. In addition, to answer RQ4, we searched and discussed the most relevant metric. Finally, we show a comparison between the selected metric and the accuracy with the different papers and datasets in Table 3

3.2.1. Classic

Classical purpose applications include classification, in particular image classification, natural language processing (NLP), or applications aimed at improving the process of machine learning itself.

Among the classification applications, we find in [31] both the resolution of the NARMA task [91], which is a benchmark test used to evaluate the performance of the model for temporal information processing and the classification of different objects based on the sensor

Table 3 Common datasets

common aatabetoi																
Dataset	[84]	[43]	[40]	[85]	[<mark>86</mark>]	[41]	[32]	[87]	[<mark>88</mark>]	[<mark>89</mark>]	[59]	[55]	[72]	[58]	[60]	[61]
MNIST	Х	Х	Х	Х	Х						Х	Х	Х	Х	Х	
TCIA		Х				х										
GTSRB		Х														
IRIS	Х						Х	Х						Х		
BNA							х									
WIL							х									
Plane Point								х								
Cats vs Dogs									Х							
CIFAR-10									Х	Х						
MedMNIST									Х	Х						Х
Fashion-MNIST									Х	Х				Х		



Fig. 27. Single transformer layer [61].

data obtained when grasping them. In addition, in [92], the Zidan model is presented; this model is used to solve problems based on the degree of entanglement between two auxiliary qubits. Furthermore, this computational model proposes an algorithm that distinguishes the most probable class between two class labels (0 or 1), when an incomplete input is introduced to the proposed model. Regarding sequence memorization and finding structures in a string chain in [55] RQNN are used to achieve these tasks. Finally, the model proposed in [42] allows classifying and distinguishing in Boolean AND, OR, XOR, and NAND functions.

Regarding image classification models, we find pattern recognition in images [44]; this model allows classifying if there are straight lines in images of 2×2 pixels, either horizontal or vertical lines [Fig. 28]. The same author also presents in [35] a classification model for 4×4 pixel images [Fig. 29].

On the other hand, classification models which are trained with well-known datasets such as the MNIST [93], images of handwritten digits from 0 to 9, the IRIS dataset [94] that represents by four features three species of the Iris flower, the Cancer Imaging Archive (TCIA) [95] hosts a large archive of medical images of cancer, the German Traffic Sign Recognition Benchmark (GTSRB) [96] contains 43 classes of traffic signs, the Banknote Authentication (BNA) [97] includes images taken from forged banknote-like specimens. In addition, the Wireless Indoor Localization dataset (WIL) [98] holds wifi signal strength observed on a smartphone, the Plane Point dataset [87] contains two types of circles, the Cats vs. Dogs dataset [99] includes different breeds of cats and dogs, CIFAR-10 [100] contains a collection of images commonly used in machine learning like ten different classes represent airplanes, cars, birds, dogs, horses, ships, etc. Other datasets used in the reviewed papers are MedMNIST, a collection of 12 pre-processed open medical image datasets [101], and Fashion-MNIST [102] is a group of 28×28 grayscale images, associated with a label from 10 classes.

In order to measure the success rate of the algorithms, the accuracy metric is used in the majority of the papers; this metric is useful because it is possible to compare it with the results obtained using classical



Fig. 28. Classification of 2×2 pixel images [44].

algorithms in other articles, given the open access to the different datasets. In the following tables, we show the results of the accuracy metrics obtained for these datasets (see Table 4).

In addition, other classical applications are the implementation of a generic mathematical model that takes advantage of quantum parallelism to speed up machine learning algorithms [103]; this model is applied in high dimensional vector spaces or the implementation of an architecture to simulate the calculation of an arbitrary non-linear function [104]. Furthermore, an example of training using OpenAI Gym, a toolkit for developing and comparing machine learning algorithms, is to solve the frozen lake game, where the player moves across a board with tiles, and the model must learn not to fall into the tiles with water on them [105]. Also, we can find a model able to forecast 24-hour-ahead wind speeds in [51] and articles about high entropy alloys [106]. Among the applications related to the physics of the Large



Fig. 29. Classification of 4×4 pixel images [44].

Table 4 Datasets metrics

Article	Dataset	Training precision (%)	Testing precision (%)
[84]	MNIST	83,36	
[43]	MNIST	99,16	98,97
[40]	MNIST	98,40	
[85]	MNIST	91,10	91,70
[59]	MNIST		98,20
[58]	MNIST	93,60	87,50
[<mark>60</mark>]	MNIST	99,69	
[58]	Fashion-MNIST	97,66	93,16
[43]	GTSRB	92,56	91,40
[41]	TCIA	99,60	
[32]	IRIS	87,50	92,10
[87]	IRIS	93,33	96,66
[32]	BNA	88,53	89,50
[32]	WIL	89,60	91,73
[87]	Plane Point	97,07	96,82
[88]	Cats vs Dogs	98,70	
[88]	CIFAR-10	94,05	
[61]	PathMNIST		75,50
[61]	ChestMNIST		94,80
[61]	DermaMNIST		72,70
[61]	OCTMNIST		60,80
[61]	PneumoniaMNIST		90,20
[61]	RetinaMNIST		54,80
[61]	BreastMNIST		83,30
[61]	BloodMNIST		88,80
[<mark>61</mark>]	TissueMNIST		59,60
[<mark>61</mark>]	OrganAMNIST		77,00
[<mark>61</mark>]	OrganCMNIST		78,70
[61]	OrganSMNIST		62,00

Hadron Collider (LHC), we find a study that applies a quantum support vector machine (QSVM-Kernel) to try to probe the coupling of the Higgs boson to the top quark, the heaviest known fermion [33].

To conclude, we elaborate a classification of real-world problems and papers, which the structure presented in [107] (see Table 5).

3.2.2. Quantum

The quantum applications include areas related to quantum circuits like data encoding, compilers or the detection of entanglement, identification of faulty gates, and quantum machine learning. The last one has subareas such as feature reduction or quantum natural language processing.

In [113], the data encodings for binary quantum classification and their properties in measurements obtained with and without noise are studied. In this data preparation stage, the authors in [114] proposed an NN called DeepQPrep, that is able to generate quantum circuits for state preparation. Another quantum circuit application is a compiler that can replace the Qiskit compiler, which uses low-depth, IBM Quantum-compatible quantum circuits that generate Greenberger-Horne-Zeilinger (GHZ) interleaved states [115]. Regarding quantum resource reduction, a sorting method designed for quantum processing units that allow working with a smaller number of quantum resources but without any decrease in efficiency is presented in [116]. This is achieved through a classification protocol with the information distributed among several agents. In addition, in [117], try to construct streamlined circuits using minimal qubits and entanglement gates to evade barren plateau phenomena and reduce computational times. Finally, in [118], discrete-variable and continuous-variable quantum neural networks are used to detect entanglement in different circuits. Regarding the error correction problem, the authors in [119] propose an ML algorithm using a recurrent neural network to identify bit-flip errors from continuous noisy syndrome measurements. Moreover, we find techniques that attempt to clone quantum states. Since the nocloning theorem states that creating an identical copy of an arbitrary unknown quantum state is impossible, techniques are being developed to clone unknown states with high fidelities rather than exact copies. For example, in [112], the authors attempted to clone an unknown state in IBM's QASM simulator using a quantum reinforcement learning protocol instead of the usual cloning method, quantum tomography. Finally, the reconstruction of the distribution of measurement results in a quantum circuit is presented in [120]; the authors used a variational autoencoder (VAE). Also, they determined that states with higher entanglement are more challenging to learn, and due to current implementations of quantum logic gates can be very faulty and introduce errors, a model is presented that identifies such faulty gates [37]. Concerning the task of finding an interpretation or explanation for quantum circuits, the authors in [62] assign a Shapley Value [121,122] to every gate of the circuit and consider that a value function that is determined by the measurement results. With this method, they are able to evaluate the quality of different ansatz.

On the other hand, we have quantum machine learning applications, like the gradient calculation in neural networks in circuits with noise [123], whose authors find that for the computation of the required gradient in a noisy device, a quantum circuit with a large number of parameters is disadvantageous. This limitation is overcome by introducing a smaller circuit, and the QNN performs well at current quantum hardware noise levels. Also, it is shown that a model with fewer parameters present in the output classifiers of quantum generative adversarial networks (GANs) facilitates discrimination. As well as that, noise in quantum devices increases the overlap between states as the gates of the circuit are applied. In addition, in [78], the authors proposed a transfer learning method between classical-quantum models, quantum–classical models, and quantum-quantum models. Moreover,

D. Peral-García et al.

Table 5

[00] [0]

[00]

[112]

T	v	17	17	17	17	37			v					
Algorithm	[44]	[35]	[84]	[43]	[<mark>40</mark>]	[85]	[32]	[58]	[108]					
Relationship between paper and solved problem.														

ngonum	[]	[33]	[04]	[43]		[00]	[32]	[30]	[100]	[0]	[04]	[07]	[107]	[110]	[111]	[112]	[)0]	[0]	[33]
Image classification	Х	Х	Х	х	Х	Х	х	Х	Х	Х	Х	Х							
Natural language processing																			
Reinforcement learning														Х	Х	Х			
Wireless communication																	Х	Х	
Collider phenomenology																			Х

[00]

[0/1

[97]

[100]

[110]

[111]

in [124], a new quantum feature subset selection technique is presented that helps to reduce the dimension of a dataset; this is achieved using quantum parallel amplitude estimation and quantum amplitude amplification techniques. Finally, we find the conceptual and mathematical foundations for quantum natural language processing (QNLP) are provided in [110] and a benchmark to evaluate the performance of matrix product state (MPS) in quantum assisted machine learning (QAML) models [86].

Another important topic is Quantum Error Correction, in [125,126] compares different quantum error correction techniques in Quantum state tomography (QST), for example, Measurement error mitigation (MEM) [26] which can mitigate the errors of a noisy measurement device, Zero-noise extrapolation (ZNE) which extrapolates the noisefree expectation value of an observable M [19], Least square error mitigation (LSEM) [28] that estimates the expectation of M at multiple random noise levels and the application of Neural network models for reduce state preparation and measurements (SPAM) errors with a neural network (NN) [14]. Also, approach based on Coherent Pauli Checks (CPCs) [127] is able detect errors in a Clifford circuit by verifying commutation rules between random Pauli-type check operators and the considered circuit and in [128] the authors proposed Variational Approach to Quantum Error Mitigation (VAQEM), which dynamically tailors existing error mitigation techniques to the VQAs by tuning specific features of these mitigation techniques. In [129], authors try to mitigate the impact of the quantum gate noise on the trainability of parameterized quantum circuits (POCs). The authors analyze different impacts like shot noise or gate noise and introduce a quasi-probabilistic error mitigation (QEM) for evaluating the stochastic gradient estimator on the noisy PQC. To conclude, there exist metrics for the measurement of error mitigation; in [130], they combine some mitigation error methods such as measurement error mitigation (MEM), randomized compiling (RC), zero noise extrapolation (ZNE), and dynamical decoupling (DD) for the metric creation.

3.3. Devices

In the case of RQ5, we introduce the different hardware environments used in the papers and their particularities. Implementing quantum devices is necessary because the same quantum circuit with different implementations can produce different results. For example, depending on the implementation, the relaxation time (T1), the time it takes for the qubit to decay from state $|1\rangle$ to $|0\rangle$, or the decoherence time (T2), the time at which the qubit loses its coherence, are different. Other factors that can be affected are the state preparation, the efficiency in applying gates, and the readout results.

One of the first companies to develop quantum devices was IBM. Since 2016, it has offered open access to its experimental quantum computing platform, called IBM Quantum. All the devices are based on superconducting qubits that are insensitive to charge noise and have a decoherence time of $\simeq 100 \ \mu s$ [115]. IBM Quantum https://quantum-computing.ibm.com/ is an experimental platform for testing quantum algorithms. A visual web tool allows for composing quantum circuits and running simulated or real experiments. Moreover, circuits can be designed using the OpenQASM language [131], and experiments can be defined and run using the Qiskit Python library [132]. Not all physical gates are implemented directly in the devices, but a transpilation process takes place to convert the programmed gates into physical gates.

Rigetti systems are powered by superconducting qubit-based quantum processors, in which qubits are coupled to a linear superconducting resonator for readout. Their principal advantages are fast gate times and fast program execution times, making them ideally suited to NISQera applications and the requirements for quantum error correction and fault-tolerant quantum computing. Among the tools offered by Rigetti are PyQuil [133] as a Python library, which allows both programming and compiling quantum circuits. To compile these circuits, the Quilc https://github.com/quil-lang/quilc compiler is used. This compiler accepts not only PyQuil as input but also Qiskit, Cirq [134] or OpenQASM language and also allows execution on non-Rigetti quantum processors. For the simulation of the execution of these circuits, Rigetti provides access to the Quantum Virtual Machine QVM https://pyquil-docs.rigetti. com/en/1.9/qvm.html, the open-source implementation of a quantum computing simulator.

TensorFlow Quantum (TFQ) [135] is a quantum machine learning library used for prototyping classical-quantum hybrid models. It uses the Cirq language, and these models are compatible with existing TensorFlow APIs. In addition, high-performance quantum circuit simulators are provided to run them. Amazon Braket https://aws.amazon. com/es/braket/ provides access to quantum processing unit (QPU) devices from different vendors such as D-Wave https://www.dwavesys. com/, IonQ https://ionq.com/ and Rigetti https://www.rigetti.com/, as well as three proprietary simulators (SV1 DM1 and TN1). Matlab https://matlab.mathworks.com/ provides a library that implements the simulation of a universal quantum computer. It allows the user to simulate quantum algorithms before implementing them on real quantum computers.

In [26] a review of different platforms that offer quantum simulation services, such as the IBM Quantum's simulator, called *ibmq_qasm_simulator* [33,43,108] in various articles, as well as the use of different physical devices such as *ibmqx2* [136–138], *ibmqx4* [138], *ibmq 16 melbourne* [31,32,136], *ibmq_rome* [108], *ibmq_santiago* [108], *ibmq_paris* [33] and *ibmq_toronto* [31] or Rigetti devices [49,112,139], Google Tensorflow Quantum Simulator [33], Amazon Braket Local Simulator [33] or the Matlab simulator [42,140].

4. Discussion

Among the algorithms used, most of them used the qubit as a minimum unit to use them in the circuit definition and implementation. However, an alternative is shown in [41] where the qutrit is presented as a helpful alternative in specific tasks. Moreover, we observe many techniques derived from neural networks, such as convolutional, orthogonal, feedback, deep, and autoencoders. The use of quantum neurons and the quantum circuit implemented - designed using rotational gates and auxiliary qubits to perform operations - are the standard features in the different varieties of QNNs. This development of quantum neural networks stems from the fact that they allow their correct execution and can provide a quantum advantage in noisy quantum computers. However, other algorithms like the QSVM give rise to a 2^N dimensional feature space (N is the number of qubits), which potential will be achieved after the NISQ era. In contrast to the neural networks, we detect few articles in other disciplines of machine learning, like natural language processing or linear regression. The articles found in these two areas establish the basis to define and implement these new emerging fields. Also, most of these machine learning models

are hybrid implementations, i.e., the models consist of both classical and quantum layers. In summary, hybrid implementations of classical machine learning models appear, such as the k-nearest neighbor model or support vector machines. The quantum algorithm is applied in a specific task of the process, i.e., calculating the distances between the vectors and finding the separation of the hyperplane, respectively. We find many hybrid implementations where the quantum algorithms are applied to improve a specific step of the classical process. Moreover, a review of the Variational Quantum Eigensolver (VQE) is done in [63]. This algorithm is used to model complex wavefunctions in polynomial time and present some degree of resilience to the noise in the quantum hardware. However, a large number of measurements are required to perform the optimization process.

In classical applications, one of the main lines of research is focused on classification. Although some papers focus on label classification, most of them focus on image classification. Image classification algorithms use known datasets such as MNIST or IRIS, or 2 × 2 or 4 × 4 pixel images, consisting of a reduced number of categories to differentiate (due to the low number of qubits currently available in quantum devices). Even if these problems are already solved in classical machine learning, they achieve comparable results in terms of accuracy. Also, [124] shows a comparison in terms of CPU computational time, and the hybrid approach can complete the training process in a shorter time than other classical approaches. However, some articles already try to take advantage of quantum advantages to achieve results that are not achievable with classical algorithms, such as the use of quantum parallelism to improve the speed of machine learning algorithms [103]. The latter articles are necessary to demonstrate the quantum advantage over classical computing. On the other hand, there are also newly discovered fields like QNLP, such as [110], where the authors establish a possible basis for further development and implementation.

In quantum applications, we notice a large number of articles that aim to improve the process of creating quantum circuits. These articles cover all the processes of creation of the ansatz itself; we highlight [116] an article that can allow NISQ devices, i.e., devices with current technology, to achieve outstanding results by being able to use a small number of quantum resources. Another relevant article is the one that challenges the no-cloning rule in [112] without using quantum tomography. Regarding the specific applications of quantum machine learning, the aim is to improve the process of creating neural networks. The creation of models with quantum applications is driven by the need to improve current noisy quantum computers and their physical limitations. In addition, the authors in [141] created a platform named Qsun, where a wavefunction represents a quantum register, quantum gates are manipulated directly by updating the amplitude of the wavefunction, and the measurement results rely on probabilities of the wavefunction. Also, this platform has QML algorithms implemented, like quantum linear regression or quantum neural networks.

In terms of devices, we find a wide variety of quantum simulators on different platforms, which allow testing as if they were real quantum devices and even allow the noise of these devices to be simulated. So, the current hardware capacity does not hold back the potential for algorithm and software development. We also find that most platforms offer support or allow translation for the OpenQASM language, which is universal, allowing research and implementation without restrictions on specific private technologies. Quantum devices today do not present a similarity in their hardware; we can observe both devices made with superconductors (IBM or Rigetti) and devices with trapped ions (IonQ). In addition, devices are made with topological cubits, photons, NV center (diamond), neutral atoms, or quantum dots. Each company opts for one or more technologies when it comes to physically implementing the devices. Each one has its advantages and disadvantages, there is no consensus or predominance of one in the literature. Today, access to quantum devices for regular users is provided by large companies through cloud computing with limitations. However, everyday access

to them is not yet possible due to the size of the devices, the complexity of implementation, and constant improvement.

To conclude, we want to note some brief comments about the potential threats to validity for this study. To mitigate the bias in certain aspects in the paper, we applied some techniques such as the quality assurance checklist [22]. Also, to reduce the bias triggered by the number of researchers involved in selecting papers, the first author was the primary reviewer. The other two authors validated the steps of the process. In addition, the public Git repository that contained the results from databases and other resources used in this research can be consulted to reproduce the process. Regarding the databases used, all the central relevant digital libraries and databases in the field of computer science were included. However, we introduce IBM Quantum Network - a not indexed ensured contrasted quality content - because it contains essential articles not included in the other databases. To ensure the validity of the results and reduce the trustiness of the discussion and conclusions achieved, we have been as rigorous as possible with the process and methodology. For this reason, we have used general terms to perform the query, well-known inclusion and exclusion criteria, and we make available the traceability of the whole process. Furthermore, all the authors have reviewed all the materials and are publicly available. So, we believe that our results deserve enough confidence due to the rigorous methodology process.

5. Conclusions

This paper presents a systematic review to identify, classify, and analyze quantum machine learning algorithms and their applications. For this, all the publications related to this knowledge area were examined unbiasedly. During the review process of the papers published between 2017 and 2022, we initially found 5497 papers from 5 different databases. Then, we reduced this amount by removing duplicates and using different criteria and quality assessments. Finally, we got the 94 papers deemed the most relevant for this research.

Considering the content of the papers, we detected different quantum machine learning designs and implementations. The main trends were related to the neural networks, and we find various quantum networks like the orthogonal, convolutional, feed-forward, or selfsupervised. Also noteworthy is the use of linear regressions, quantum amplitude estimation (QAE), variational depth quantum circuit (vVQC), quantum Boltzmann machines (QBM). Regarding classical applications, a large number of papers focus on image classification with well-known datasets such as MNIST. On the other hand, in quantum applications, many of them focus on improving the process of creating, simulating, and executing ansatzs.

During our literature review, we detected the existence of standardization in the encoding of the data and the initialization of the states, but many different implementations and proposals in the creation of oracles and quantum circuits in general. Nevertheless, classical applications try to emulate the problems already solved by classical machine learning, obtaining, in the best case, the same result as the existing classical models.

An improvement in the quantum hardware will support the improvement of the results of the applied techniques. Because many of the applied methods, such as image classification, are limited by the different possible output labels, which are proportional to the number of qubits in the device.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Data will be made available on request.

D. Peral-García et al.

References

- [1] A. Zulehner, R. Wille, Simulation and design of quantum circuits, in: I. Ulidowski, I. Lanese, U.P. Schultz, C. Ferreira (Eds.), Reversible Computation: Extending Horizons of Computing: Selected Results of the COST Action IC1405, Springer International Publishing, Cham, 2020, pp. 60–82, http://dx.doi.org/10. 1007/978-3-030-47361-7_3.
- [2] M. Benedetti, E. Lloyd, S. Sack, M. Fiorentini, Parameterized quantum circuits as machine learning models, Quantum Sci. Technol. 4 (4) (2019) http://dx.doi. org/10.1088/2058-9565/ab4eb5, arXiv:1906.07682.
- [3] L.K. Grover, A fast quantum mechanical algorithm for database search, in: Proceedings of the Twenty-Eighth Annual ACM Symposium on Theory of Computing, STOC '96, Association for Computing Machinery, New York, NY, USA, 1996, pp. 212–219, http://dx.doi.org/10.1145/237814.237866.
- [4] P.W. Shor, Polynomial-time algorithms for prime factorization and discrete logarithms on a quantum computer, SIAM J. Comput. 26 (5) (1997) 1484–1509, http://dx.doi.org/10.1137/s0097539795293172.
- [5] Y. Cao, J. Romero, J.P. Olson, M. Degroote, P.D. Johnson, M. Kieferová, I.D. Kivlichan, T. Menke, B. Peropadre, N.P.D. Sawaya, S. Sim, L. Veis, A. Aspuru-Guzik, Quantum chemistry in the age of quantum computing, Chem. Rev. 119 (19) (2019) 10856–10915, http://dx.doi.org/10.1021/acs.chemrev.8b00803, PMID: 31469277. arXiv:https://doi.org/10.1021/acs.chemrev.8b00803.
- [6] S. McArdle, S. Endo, A. Aspuru-Guzik, S.C. Benjamin, X. Yuan, Quantum computational chemistry, Rev. Modern Phys. 92 (2020) 015003, http:// dx.doi.org/10.1103/RevModPhys.92.015003, URL https://link.aps.org/doi/10. 1103/RevModPhys.92.015003.
- [7] B. Bauer, S. Bravyi, M. Motta, G.K.-L. Chan, Quantum algorithms for quantum chemistry and quantum materials science, Chem. Rev. 120 (22) (2020) 12685–12717, http://dx.doi.org/10.1021/acs.chemrev.9b00829, PMID: 33090772. arXiv:https://doi.org/10.1021/acs.chemrev.9b00829.
- [8] M. Ashwin, A.S. Alqahtani, A. Mubarakali, B. Sivakumar, Efficient resource management in 6G communication networks using hybrid quantum deep learning model, Comput. Electr. Eng. 106 (2023) 108565, http://dx.doi. org/10.1016/j.compeleceng.2022.108565, URL https://www.sciencedirect.com/ science/article/pii/S0045790622007807.
- [9] A.W. Harrow, A. Hassidim, S. Lloyd, Quantum algorithm for linear systems of equations, Phys. Rev. Lett. 103 (2009) 150502, http://dx.doi.org/10.1103/ PhysRevLett.103.150502, URL https://link.aps.org/doi/10.1103/PhysRevLett. 103.150502.
- [10] B. Nachman, D. Provasoli, W.A. de Jong, C.W. Bauer, Quantum algorithm for high energy physics simulations, Phys. Rev. Lett. 126 (6) (2021) http: //dx.doi.org/10.1103/physrevlett.126.062001.
- [11] C.W. Bauer, M. Freytsis, B. Nachman, Simulating collider physics on quantum computers using effective field theories, 2021, arXiv:2102.05044.
- [12] C.H. Bennett, G. Brassard, Quantum cryptography: Public key distribution and coin tossing, Theoret. Comput. Sci. 560 (2014) 7–11, http://dx.doi.org/10. 1016/j.tcs.2014.05.025.
- [13] V. Scarani, H. Bechmann-Pasquinucci, N.J. Cerf, M. Dušek, N. Lütkenhaus, M. Peev, The security of practical quantum key distribution, Rev. Modern Phys. 81 (3) (2009) 1301–1350, http://dx.doi.org/10.1103/revmodphys.81.1301.
- [14] W. Zhang, D.-S. Ding, Y.-B. Sheng, L. Zhou, B.-S. Shi, G.-C. Guo, Quantum secure direct communication with quantum memory, Phys. Rev. Lett. 118 (22) (2017) http://dx.doi.org/10.1103/physrevlett.118.220501.
- [15] J. Biamonte, P. Wittek, N. Pancotti, P. Rebentrost, N. Wiebe, S. Lloyd, Quantum machine learning, Nature 549 (7671) (2017) 195–202, http://dx.doi.org/10. 1038/nature23474.
- [16] M. Schuld, I. Sinayskiy, F. Petruccione, An introduction to quantum machine learning, Contemp. Phys. 56 (2) (2014) 172–185, http://dx.doi.org/10.1080/ 00107514.2014.964942.
- [17] Y. Chen, Y. Pan, G. Zhang, S. Cheng, Detecting quantum entanglement with unsupervised learning, 2021, arXiv:2103.04804.
- [18] M.-Z. Ai, Y. Ding, Y. Ban, J.D. Martín-Guerrero, J. Casanova, J.-M. Cui, Y.-F. Huang, X. Chen, C.-F. Li, G.-C. Guo, Experimentally realizing efficient quantum control with reinforcement learning, 2021, arXiv:2101.09020.
- [19] L. Domingo, F. Borondo, Deep learning methods for the computation of vibrational wavefunctions, 2021, arXiv:2103.00202.
- [20] D. Alaminos, M.B. Salas, M.A. Fernández-Gámez, Quantum computing and deep learning methods for GDP growth forecasting, Comput. Econ. (2021) http://dx.doi.org/10.1007/s10614-021-10110-z.
- [21] S. Jain, J. Ziauddin, P. Leonchyk, S. Yenkanchi, J. Geraci, Quantum and classical machine learning for the classification of non-small-cell lung cancer patients, SN Appl. Sci. 2 (6) (2020) 1088, http://dx.doi.org/10.1007/s42452-020-2847-4.
- [22] B.A. Kitchenham, S. Charters, Guidelines for Performing Systematic Literature Reviews in Software Engineering, Tech. Rep. EBSE 2007-001, Keele University and Durham University Joint Report, 2007, URL https://www.elsevier.com/_ data/promis_misc/525444systematicreviewsguide.pdf.

- [23] J. Cruz-Benito, F.J. García-Peñalvo, R. Therón, Analyzing the software architectures supporting HCI/HMI processes through a systematic review of the literature, Telemat. Inform. 38 (2019) 118–132, http://dx.doi.org/10. 1016/j.tele.2018.09.006, URL http://www.sciencedirect.com/science/article/ pii/S0736585318305392.
- [24] A.G. Holgado, S.M. Pablos, F.J.G. Peñalvo, Guidelines for performing systematic research projects reviews, Int. J. Interact. Multimed. Artif. Intell. 6 (2) (2020) 136–144, http://dx.doi.org/10.9781/ijimai.2020.05.005, URL http: //hdl.handle.net/10366/143081.
- [25] F.J. García-Peñalvo, Desarrollo de estados de la cuestión robustos: Revisiones sistemáticas de literatura, Educ. Knowl. Soc. (EKS) 23 (2022) http://dx.doi.org/ 10.14201/eks.28600, URL http://repositorio.grial.eu/handle/grial/2568.
- [26] W. O'Quinn, S. Mao, Quantum machine learning: Recent advances and outlook, IEEE Wirel. Commun. 27 (3) (2020) 126–131, http://dx.doi.org/10.1109/MWC. 001.1900341.
- [27] D. Moher, A. Liberati, J. Tetzlaff, D.G. Altman, Preferred reporting items for systematic reviews and meta-analyses: The PRISMA statement, Int. J. Surg. 8 (5) (2010) 336–341, http://dx.doi.org/10.1016/j.ijsu.2010.02.007, URL https: //www.sciencedirect.com/science/article/pii/S1743919110000403.
- [28] M. Petticrew, H. Roberts, Systematic Reviews in the Social Sciences: A Practical Guide, vol. 11, 2006, http://dx.doi.org/10.1002/9780470754887.
- [29] Y. Huang, H. Lei, X. Li, Q. Zhu, W. Ren, X. Liu, Quantum generative model with variable-depth circuit, Comput. Mater. Contin. 65 (1) (2020) 445–458, http://dx.doi.org/10.32604/cmc.2020.010390.
- [30] M. Srikumar, C.D. Hill, L.C.L. Hollenberg, Clustering and enhanced classification using a hybrid quantum autoencoder, Quantum Sci. Technol. 7 (1) (2021) 015020, http://dx.doi.org/10.1088/2058-9565/ac3c53.
- [31] Y. Suzuki, Q. Gao, K. Pradel, K. Yasuoka, N. Yamamoto, Natural quantum reservoir computing for temporal information processing, 2021, arXiv:2107. 05808.
- [32] A. Chalumuri, R. Kune, B.S. Manoj, A hybrid classical-quantum approach for multi-class classification, Quantum Inf. Process. 20 (3) (2021) http://dx.doi. org/10.1007/s11128-021-03029-9.
- [33] S.L. Wu, S. Sun, W. Guan, C. Zhou, J. Chan, C.L. Cheng, T. Pham, Y. Qian, A.Z. Wang, R. Zhang, M. Livny, J. Glick, P.K. Barkoutsos, S. Woerner, I. Tavernelli, F. Carminati, A.D. Meglio, A.C.Y. Li, J. Lykken, P. Spentzouris, S.Y.-C. Chen, S. Yoo, T.-C. Wei, Application of quantum machine learning using the quantum kernel algorithm on high energy physics analysis at the LHC, 2021, arXiv:2104.05059.
- [34] B.Q. Chen, X.F. Niu, Quantum neural network with improved quantum learning algorithm, Internat. J. Theoret. Phys. 59 (7) (2020) 1978–1991, http://dx.doi. org/10.1007/s10773-020-04470-9.
- [35] F. Tacchino, S. Mangini, P.K. Barkoutsos, C. Macchiavello, D. Gerace, I. Tavernelli, D. Bajoni, Variational learning for quantum artificial neural networks, IEEE Trans. Quantum Eng. 2 (2021) 1–10, http://dx.doi.org/10.1109/tqe.2021. 3062494.
- [36] J. Bausch, Classifying data using near-term quantum devices, Int. J. Quantum Inf. 16 (8) (2018) http://dx.doi.org/10.1142/S0219749918400014.
- [37] M.L. LaBorde, A.C. Rogers, J.P. Dowling, Finding broken gates in quantum circuits: exploiting hybrid machine learning, Quantum Inf. Process. 19 (8) (2020) http://dx.doi.org/10.1007/s11128-020-02729-y, arXiv:2001.10939.
- [38] K. Jia, S. Li, Y. Wen, T. Liu, D. Tao, Orthogonal deep neural networks, 2019, arXiv:1905.05929.
- [39] J. Wang, Y. Chen, R. Chakraborty, S.X. Yu, Orthogonal convolutional neural networks, 2020, arXiv:1911.12207.
- [40] I. Kerenidis, J. Landman, N. Mathur, Classical and quantum algorithms for orthogonal neural networks, 2021, arXiv:2106.07198.
- [41] D. Konar, S. Bhattacharyya, B.K. Panigrahi, E.C. Behrman, Qutrit-inspired fully self-supervised shallow quantum learning network for brain tumor segmentation, IEEE Trans. Neural Netw. Learn. Syst. (2021) 1–15, http://dx.doi.org/10. 1109/tnnls.2021.3077188, arXiv:2009.06767.
- [42] M. Lukac, K. Abdiyeva, M. Kameyama, CNOT-measure quantum neural networks, in: Proceedings of the International Symposium on Multiple-Valued Logic, Vol. 2018-May, IEEE Computer Society, 2018, pp. 186–191, http://dx. doi.org/10.1109/ISMVL.2018.00040.
- [43] Y. Li, R.G. Zhou, R. Xu, J. Luo, W. Hu, A quantum deep convolutional neural network for image recognition, Quantum Sci. Technol. 5 (4) (2020) http://dx.doi.org/10.1088/2058-9565/ab9f93.
- [44] F. Tacchino, P. Barkoutsos, C. Macchiavello, I. Tavernelli, D. Gerace, D. Bajoni, Quantum implementation of an artificial feed-forward neural network, Quantum Sci. Technol. 5 (4) (2020) http://dx.doi.org/10.1088/2058-9565/abb8e4, arXiv: 1912.12486.
- [45] H. Wang, J. Zhao, B. Wang, L. Tong, A quantum approximate optimization algorithm with metalearning for maxcut problem and its simulation via tensorflow quantum, Math. Probl. Eng. 2021 (2021) http://dx.doi.org/10.1155/2021/ 6655455.
- [46] C.P. Gonçalves, Quantum neural machine learning: Backpropagation and dynamics, NeuroQuantology 15 (1) (2017) 22–41, http://dx.doi.org/10.14704/ nq.2017.15.1.1008, arXiv:1609.06935.

- [47] H. Situ, Z. He, Y. Wang, L. Li, S. Zheng, Quantum generative adversarial network for generating discrete distribution, Inform. Sci. 538 (2020) 193–208, http://dx.doi.org/10.1016/j.ins.2020.05.127, arXiv:1807.01235.
- [48] W. Liu, Y. Zhang, Z. Deng, J. Zhao, L. Tong, A hybrid quantum-classical conditional generative adversarial network algorithm for human-centered paradigm in cloud, Eurasip J. Wirel. Commun. Netw. 2021 (1) (2021) http://dx.doi.org/ 10.1186/s13638-021-01898-3.
- [49] A. Anand, J. Romero, M. Degroote, A. Aspuru-Guzik, Noise robustness and experimental demonstration of a quantum generative adversarial network for continuous distributions, Adv. Quantum Technol. (2021) http://dx.doi.org/10. 1002/qute.202000069, arXiv:2006.01976.
- [50] A. Ceschini, A. Rosato, M. Panella, Design of an LSTM cell on a quantum hardware, IEEE Trans. Circuits Syst. II 69 (3) (2022) 1822–1826, http://dx. doi.org/10.1109/TCSII.2021.3126204.
- [51] Y.-Y. Hong, C.J.E. Arce, T.-W. Huang, A robust hybrid classical and quantum model for short-term wind speed forecasting, IEEE Access 11 (2023) 90811–90824, http://dx.doi.org/10.1109/ACCESS.2023.3308053.
- [52] S.Y.-C. Chen, Asynchronous training of quantum reinforcement learning, Procedia Comput. Sci. 222 (2023) 321–330, http://dx.doi.org/10.1016/j.procs.2023. 08.171, International Neural Network Society Workshop on Deep Learning Innovations and Applications (INNS DLIA 2023). URL https://www.sciencedirect. com/science/article/pii/S1877050923009365.
- [53] J. Preskill, Quantum Computing in the NISQ era and beyond, Quantum 2 (2018) 79, http://dx.doi.org/10.22331/q-2018-08-06-79.
- [54] L. Gyongyosi, S. Imre, Training optimization for gate-model quantum neural networks, Sci. Rep. 9 (1) (2019) http://dx.doi.org/10.1038/s41598-019-48892w, arXiv:1909.01048.
- [55] J. Bausch, Recurrent quantum neural networks, 2020, arXiv:2006.14619.
- [56] Y. Chen, F. Li, J. Wang, B. Tang, X. Zhou, Quantum recurrent encoderdecoder neural network for performance trend prediction of rotating machinery, Knowl-Based Syst. 197 (2020) 105863, http://dx.doi.org/10.1016/ j.knosys.2020.105863, URL https://www.sciencedirect.com/science/article/pii/ S0950705120302252.
- [57] U. Ullah, A.G.O. Jurado, I.D. Gonzalez, B. Garcia-Zapirain, A fully connected quantum convolutional neural network for classifying ischemic cardiopathy, IEEE Access 10 (2022) 134592–134605, http://dx.doi.org/10.1109/ACCESS. 2022.3232307.
- [58] M. Alam, S. Ghosh, QNet: A scalable and noise-resilient quantum neural network architecture for noisy intermediate-scale quantum computers, Front. Phys. 9 (2022) http://dx.doi.org/10.3389/fphy.2021.755139.
- [59] Y. Du, Y. Qian, X. Wu, D. Tao, A distributed learning scheme for variational quantum algorithms, IEEE Trans. Quantum Eng. 3 (2022) 1–16, http://dx.doi. org/10.1109/TQE.2022.3175267.
- [60] R. Sharma, B. Kaushik, N.K. Gondhi, M. Tahir, M.K.I. Rahmani, Quantum particle swarm optimization based convolutional neural network for handwritten script recognition, Comput. Mater. Contin. 71 (3) (2022) 5855–5873, http: //dx.doi.org/10.32604/cmc.2022.024232, URL http://www.techscience.com/ cmc/v71n3/46543.
- [61] E.A. Cherrat, I. Kerenidis, N. Mathur, J. Landman, M. Strahm, Y.Y. Li, Quantum vision transformers, 2022, arXiv:2209.08167.
- [62] R. Heese, T. Gerlach, S. Mücke, S. Müller, M. Jakobs, N. Piatkowski, Explaining quantum circuits with Shapley values: Towards explainable quantum machine learning, 2023, arXiv:2301.09138.
- [63] J. Tilly, H. Chen, S. Cao, D. Picozzi, K. Setia, Y. Li, E. Grant, L. Wossnig, I. Rungger, G.H. Booth, J. Tennyson, The variational quantum eigensolver: A review of methods and best practices, Phys. Rep. 986 (2022) 1–128, http: //dx.doi.org/10.1016/j.physrep.2022.08.003.
- [64] C. Zoufal, A. Lucchi, S. Woerner, Variational quantum Boltzmann machines, Quantum Mach. Intell. 3 (1) (2021) http://dx.doi.org/10.1007/s42484-020-00033-7.
- [65] D.N. Diep, Some quantum neural networks, Internat. J. Theoret. Phys. 59 (4) (2020) 1179–1187, http://dx.doi.org/10.1007/s10773-020-04397-1.
- [66] V. Skavysh, S. Priazhkina, D. Guala, T.R. Bromley, Quantum monte carlo for economics: Stress testing and macroeconomic deep learning, J. Econom. Dynam. Control 153 (2023) 104680, http://dx.doi.org/10.1016/j.jedc.2023.104680, URL https://www.sciencedirect.com/science/article/pii/S0165188923000866.
- [67] J. Kim, J. Huh, D.K. Park, Classical-to-quantum convolutional neural network transfer learning, Neurocomputing 555 (2023) 126643, http://dx.doi.org/ 10.1016/j.neucom.2023.126643, URL https://www.sciencedirect.com/science/ article/pii/S092523122300766X.
- [68] D.H. Ackley, G.E. Hinton, T.J. Sejnowski, A learning algorithm for boltzmann machines, Cogn. Sci. 9 (1) (1985) 147–169, http://dx.doi.org/10.1016/ S0364-0213(85)80012-4, URL https://www.sciencedirect.com/science/article/ pii/S0364021385800124.
- [69] M. Benedetti, D. Garcia-Pintos, O. Perdomo, V. Leyton-Ortega, Y. Nam, A. Perdomo-Ortiz, A generative modeling approach for benchmarking and training shallow quantum circuits, npj Quantum Inf. 5 (1) (2019) http://dx.doi.org/10. 1038/s41534-019-0157-8.
- [70] M. Schuld, A. Bocharov, K.M. Svore, N. Wiebe, Circuit-centric quantum classifiers, Phys. Rev. A 101 (3) (2020) http://dx.doi.org/10.1103/physreva.101. 032308.

- [71] G. Brassard, P. Høyer, M. Mosca, A. Tapp, Quantum amplitude amplification and estimation, Quantum Comput. Inf. (2002) 53–74, http://dx.doi.org/10. 1090/conm/305/05215.
- [72] H.-Y. Huang, M. Broughton, M. Mohseni, R. Babbush, S. Boixo, H. Neven, J.R. McClean, Power of data in quantum machine learning, Nature Commun. 12 (1) (2021) http://dx.doi.org/10.1038/s41467-021-22539-9.
- [73] I.T. Jolliffe, Principal Component Analysis, 1986, p. XIII, 271, http://dx.doi. org/10.1007/978-1-4757-1904-8.
- [74] M. Schuld, A. Bocharov, K.M. Svore, N. Wiebe, Circuit-centric quantum classifiers, Phys. Rev. A 101 (3) (2020) http://dx.doi.org/10.1103/physreva.101. 032308.
- [75] M. Schuld, N. Killoran, Quantum machine learning in feature Hilbert spaces, Phys. Rev. Lett. 122 (4) (2019) http://dx.doi.org/10.1103/physrevlett.122. 040504.
- [76] A. Skolik, J.R. McClean, M. Mohseni, P. van der Smagt, M. Leib, Layerwise learning for quantum neural networks, Quantum Mach. Intell. 3 (1) (2021) http://dx.doi.org/10.1007/s42484-020-00036-4.
- [77] V. Havlíček, A.D. Córcoles, K. Temme, A.W. Harrow, A. Kandala, J.M. Chow, J.M. Gambetta, Supervised learning with quantum-enhanced feature spaces, Nature 567 (7747) (2019) 209–212, http://dx.doi.org/10.1038/s41586-019-0980-2.
- [78] A. Mari, T.R. Bromley, J. Izaac, M. Schuld, N. Killoran, Transfer learning in hybrid classical-quantum neural networks, Quantum 4 (2020) 340, http: //dx.doi.org/10.22331/q-2020-10-09-340.
- [79] D. Nicolay, T. Carletti, Quantum neural networks achieving quantum algorithms, in: Communications in Computer and Information Science, vol. 830, Springer Verlag, 2018, pp. 3–15, http://dx.doi.org/10.1007/978-3-319-78658-2_1.
- [80] A. Pérez-Salinas, A. Cervera-Lierta, E. Gil-Fuster, J.I. Latorre, Data re-uploading for a universal quantum classifier, Quantum 4 (2020) 226, http://dx.doi.org/ 10.22331/q-2020-02-06-226.
- [81] P. Festa, P. Pardalos, M. Resende, C. Ribeiro, Randomized heuristics for the Max-Cut problem, Optim. Methods Softw. 17 (6) (2002) 1033–1058, http: //dx.doi.org/10.1080/1055678021000090033, arXiv:https://doi.org/10.1080/ 1055678021000090033.
- [82] I. Goodfellow, J. Pouget-Abadie, M. Mirza, B. Xu, D. Warde-Farley, S. Ozair, A. Courville, Y. Bengio, Generative adversarial nets, in: Z. Ghahramani, M. Welling, C. Cortes, N. Lawrence, K.Q. Weinberger (Eds.), in: Advances in Neural Information Processing Systems, vol. 27, Curran Associates, Inc., 2014, URL https://proceedings.neurips.cc/paper/2014/file/ 5ca3e9b122f61f8f06494c97b1afccf3-Paper.pdf.
- [83] J.J. Moré, The Levenberg-Marquardt algorithm: Implementation and theory, in: G.A. Watson (Ed.), Numerical Analysis, Springer Berlin Heidelberg, Berlin, Heidelberg, 1978, pp. 105–116.
- [84] R. Huang, X. Tan, Q. Xu, Variational quantum tensor networks classifiers, Neurocomputing 452 (2021) 89–98, http://dx.doi.org/10.1016/j. neucom.2021.04.074, URL https://www.sciencedirect.com/science/article/pii/ S092523122100624X.
- [85] N. Thumwanit, C. Lortararprasert, H. Yano, R. Raymond, Trainable discrete feature embeddings for variational quantum classifier, 2021, arXiv:2106.09415.
- [86] M.L. Wall, M.R. Abernathy, G. Quiroz, Generative machine learning with tensor networks: Benchmarks on near-term quantum computers, Phys. Rev. Res. 3 (2) (2021) http://dx.doi.org/10.1103/physrevresearch.3.023010.
- [87] P. Li, B. Wang, Quantum neural networks model based on swap test and phase estimation, Neural Netw. 130 (2020) 152–164, http://dx.doi.org/10.1016/j. neunet.2020.07.003.
- [88] S.Y.C. Chen, S. Yoo, Federated quantum machine learning, Entropy 23 (4) (2021) http://dx.doi.org/10.3390/e23040460, arXiv:2103.12010.
- [89] T. Tomesh, P. Gokhale, E.R. Anschuetz, F.T. Chong, Coreset clustering on small quantum computers, 2020, arXiv:2004.14970.
- [90] N. Metropolis, S. Ulam, The Monte Carlo method, J. Amer. Statist. Assoc. 44 (247) (1949) 335–341, URL http://www.jstor.org/stable/2280232.
- [91] L. Appeltant, M. Soriano, G. Van der Sande, J. Danckaert, S. Massar, J. Dambre, B. Schrauwen, C. Mirasso, I. Fischer, Information processing using a single dynamical node as complex system, Nat. Commun. 2 (2011) 468, http://dx.doi.org/10.1038/ncomms1476.
- [92] A.H. Abdel-Aty, H. Kadry, M. Zidan, Y. Al-Sbou, E.A. Zanaty, M. Abdel-Aty, A quantum classification algorithm for classification incomplete patterns based on entanglement measure, J. Intell. Fuzzy Systems 38 (3) (2020) 2817–2822, http://dx.doi.org/10.3233/JIFS-179566.
- [93] Y. Lecun, L. Bottou, Y. Bengio, P. Haffner, Gradient-based learning applied to document recognition, Proc. IEEE 86 (1998) 2278–2324, http://dx.doi.org/10. 1109/5.726791.
- [94] R.A. Fisher, The use of multiple measurements in taxonomic problems, Ann. Eugen. 7 (2) (1936) 179–188, http://dx.doi.org/10.1111/j.1469-1809.1936.tb02137.x, arXiv:https://onlinelibrary.wiley.com/doi/pdf/10.1111/ j.1469-1809.1936.tb02137.x. URL https://onlinelibrary.wiley.com/doi/abs/10. 1111/j.1469-1809.1936.tb02137.x.

- [95] K. Clark, B. Vendt, K. Smith, J. Freymann, J. Kirby, P. Koppel, S. Moore, S. Phillips, D. Maffitt, M. Pringle, et al., The cancer imaging archive (TCIA): Maintaining and operating a public information repository, J. Digit. Imaging 26 (6) (2013) 1045–1057, http://dx.doi.org/10.1007/s10278-013-9622-7.
- [96] J. Stallkamp, M. Schlipsing, J. Salmen, C. Igel, Man vs. computer: Benchmarking machine learning algorithms for traffic sign recognition, Neural Netw. (2012) http://dx.doi.org/10.1016/j.neunet.2012.02.016, URL http:// www.sciencedirect.com/science/article/pii/S0893608012000457.
- [97] V. Lohweg, H. Darksen, UCI machine learning repository, 2012, URL https: //archive.ics.uci.edu/ml/datasets/banknote+authentication.
- [98] R. Bhatt, UCI machine learning repository, 2017, URL https://archive.ics.uci. edu/ml/datasets/banknote+authentication.
- [99] O.M. Parkhi, A. Vedaldi, A. Zisserman, C.V. Jawahar, Cats and dogs, in: 2012 IEEE Conference on Computer Vision and Pattern Recognition, 2012, pp. 3498–3505, http://dx.doi.org/10.1109/CVPR.2012.6248092.
- [100] A. Krizhevsky, V. Nair, G. Hinton, Cifar-10 and cifar-100 datasets, 2009, p. 1, URI: https://www.cs.toronto.edu/kriz/cifar.html, 6 (1).
- [101] J. Yang, R. Shi, B. Ni, MedMNIST classification decathlon: A lightweight AutoML benchmark for medical image analysis, in: 2021 IEEE 18th International Symposium on Biomedical Imaging, ISBI, IEEE, 2021, http://dx.doi.org/10. 1109/isbi48211.2021.9434062.
- [102] H. Xiao, K. Rasul, R. Vollgraf, Fashion-MNIST: a novel image dataset for benchmarking machine learning algorithms, 2017, arXiv:cs.LG/1708.07747.
- [103] V. Dasari, M.S. Im, L. Beshaj, Solving Machine Learning Optimization Problems Using Quantum Computers, SPIE-Intl Soc Optical Eng, 2020, p. 16, http://dx. doi.org/10.1117/12.2565038, arXiv:1911.08587.
- [104] F.M. De Paula Neto, T.B. Ludermir, W.R. De Oliveira, A.J. Da Silva, Implementing any nonlinear quantum neuron, IEEE Trans. Neural Netw. Learn. Syst. 31 (9) (2020) 3741–3746, http://dx.doi.org/10.1109/TNNLS.2019.2938899.
- [105] S.Y.C. Chen, C.H.H. Yang, J. Qi, P.Y. Chen, X. Ma, H.S. Goan, Variational quantum circuits for deep reinforcement learning, IEEE Access 8 (2020) 141007–141024, http://dx.doi.org/10.1109/ACCESS.2020.3010470, arXiv:1907.00397.
- [106] P. Brown, H. Zhuang, Quantum machine-learning phase prediction of highentropy alloys, Mater. Today 63 (2023) 18–31, http://dx.doi.org/10.1016/j. mattod.2023.02.014.
- [107] A. Jadhav, A. Rasool, M. Gyanchandani, Quantum Machine Learning: Scope for real-world problems, Procedia Comput. Sci. 218 (2023) 2612–2625, http:// dx.doi.org/10.1016/j.procs.2023.01.235, International Conference on Machine Learning and Data Engineering. URL https://www.sciencedirect.com/science/ article/pii/\$1877050923002351.
- [108] S. Yan, H. Qi, W. Cui, Nonlinear quantum neuron: A fundamental building block for quantum neural networks, Phys. Rev. A 102 (5) (2020) http://dx.doi. org/10.1103/PhysRevA.102.052421, arXiv:2011.03429.
- [109] J. Shi, Z. Li, W. Lai, F. Li, R. Shi, Y. Feng, S. Zhang, Two end-to-end quantum-inspired deep neural networks for text classification, IEEE Trans. Knowl. Data Eng. 35 (4) (2023) 4335–4345, http://dx.doi.org/10.1109/TKDE. 2021.3130598.
- [110] B. Coecke, G. de Felice, K. Meichanetzidis, A. Toumi, Foundations for near-term quantum natural language processing, 2020, arXiv:2012.03755.
- [111] Y. Chen, Y. Pan, D. Dong, Quantum language model with entanglement embedding for question answering, IEEE Trans. Cybern. 53 (6) (2023) 3467–3478, http://dx.doi.org/10.1109/TCYB.2021.3131252.
- [112] K.S. Shenoy, D.Y. Sheth, B.K. Behera, P.K. Panigrahi, Demonstration of a measurement-based adaptation protocol with quantum reinforcement learning on the IBM Q experience platform, Quantum Inf. Process. 19 (5) (2020) http://dx.doi.org/10.1007/s11128-020-02657-x.
- [113] R. Larose, B. Coyle, Robust data encodings for quantum classifiers, Phys. Rev. A 102 (3) (2020) http://dx.doi.org/10.1103/PhysRevA.102.032420, arXiv: 2003.01695.
- [114] P. Selig, N. Murphy, D. Redmond, S. Caton, DeepQPrep: Neural Network Augmented Search for Quantum State Preparation, IEEE Access 11 (2023) 76388–76402, http://dx.doi.org/10.1109/ACCESS.2023.3296802.
- [115] D. Ferrari, M. Amoretti, Efficient and effective quantum compiling for entanglement-based machine learning on IBM Q devices, Int. J. Quantum Inf. 16 (8) (2018) http://dx.doi.org/10.1142/S0219749918400063, arXiv:1801.02363.
- [116] P. Sadowski, Quantum distance-based classifier with distributed knowledge and state recycling, Int. J. Quantum Inf. 16 (8) (2018) http://dx.doi.org/10.1142/ S0219749918400130.
- [117] E. Andrés, M.P. Cuéllar, G. Navarro, Efficient dimensionality reduction strategies for quantum reinforcement learning, IEEE Access 11 (2023) 104534–104553, http://dx.doi.org/10.1109/ACCESS.2023.3318173.

- [118] P.H. Qiu, X.G. Chen, Y.W. Shi, Detecting entanglement with deep quantum neural networks, IEEE Access 7 (2019) 94310–94320, http://dx.doi.org/10. 1109/ACCESS.2019.2929084.
- [119] I. Convy, H. Liao, S. Zhang, S. Patel, W.P. Livingston, H.N. Nguyen, I. Siddiqi, K.B. Whaley, Machine learning for continuous quantum error correction on superconducting qubits, New J. Phys. 24 (6) (2022) 063019, http://dx.doi.org/ 10.1088/1367-2630/ac66f9.
- [120] H. Situ, Z. He, Machine learning distributions of quantum ansatz with hierarchical structure, Internat. J. Modern Phys. B 34 (20) (2020) http://dx.doi.org/ 10.1142/S0217979220501969.
- [121] L.S. Shapley, A Value for N-Person Games, RAND Corporation, Santa Monica, CA, 1952, http://dx.doi.org/10.7249/P0295.
- [122] R.J. Aumann, L.S. Shapley, Values of Non-Atomic Games, Princeton University Press, 1974, URL http://www.jstor.org/stable/j.ctt13x149m.
- [123] A. Patterson, H. Chen, L. Wossnig, S. Severini, D. Browne, I. Rungger, Quantum state discrimination using noisy quantum neural networks, Phys. Rev. Res. 3 (1) (2021) http://dx.doi.org/10.1103/physrevresearch.3.013063, arXiv:1911. 00352.
- [124] S. Chakraborty, S.H. Shaikh, A. Chakrabarti, R. Ghosh, A hybrid quantum feature selection algorithm using a quantum inspired graph theoretic approach, Appl. Intell. 50 (6) (2020) 1775–1793, http://dx.doi.org/10.1007/s10489-019-01604-3.
- [125] S. Ramadhani, J.U. Rehman, H. Shin, Quantum error mitigation for quantum state tomography, IEEE Access 9 (2021) 107955–107964, http://dx.doi.org/10. 1109/ACCESS.2021.3101214.
- [126] D. Qin, X. Xu, Y. Li, An overview of quantum error mitigation formulas, Chin. Phys. B 31 (9) (2022) 090306, http://dx.doi.org/10.1088/1674-1056/ac7ble.
- [127] E. van den Berg, S. Bravyi, J.M. Gambetta, P. Jurcevic, D. Maslov, K. Temme, Single-shot error mitigation by coherent Pauli checks, 2022, arXiv:2212.03937.
- [128] G.S. Ravi, K.N. Smith, P. Gokhale, A. Mari, N. Earnest, A. Javadi-Abhari, F.T. Chong, VAQEM: A variational approach to quantum error mitigation, 2021, arXiv:2112.05821.
- [129] S.T. Jose, O. Simeone, Error mitigation-aided optimization of parameterized quantum circuits: Convergence analysis, 2022, arXiv:2209.11514.
- [130] A.A. Saki, A. Katabarwa, S. Resch, G. Umbrarescu, Hypothesis testing for error mitigation: How to evaluate error mitigation, 2023, arXiv:2301.02690.
- [131] A.W. Cross, L.S. Bishop, J.A. Smolin, J.M. Gambetta, Open quantum assembly language, 2017, arXiv:1707.03429.
- [132] M.D. Sajid Anis, et al., Qiskit: An open-source framework for quantum computing, 2021, http://dx.doi.org/10.5281/zenodo.2573505.
- [133] R.S. Smith, M.J. Curtis, W.J. Zeng, A practical quantum instruction set architecture, 2016, arXiv:1608.03355.
- [134] C. Developers, Cirq, 2021, http://dx.doi.org/10.5281/zenodo.5182845, See full list of authors on Github: https://github.com/quantumlib/Cirq/graphs/ contributors.
- [135] M. Broughton, G. Verdon, T. McCourt, A.J. Martinez, J.H. Yoo, S.V. Isakov, P. Massey, R. Halavati, M.Y. Niu, A. Zlokapa, E. Peters, O. Lockwood, A. Skolik, S. Jerbi, V. Dunjko, M. Leib, M. Streif, D.V. Dollen, H. Chen, S. Cao, R. Wiersema, H.-Y. Huang, J.R. McClean, R. Babbush, S. Boixo, D. Bacon, A.K. Ho, H. Neven, M. Mohseni, TensorFlow quantum: A software framework for quantum machine learning, 2021, arXiv:2003.02989.
- [136] K. Kathuria, A. Ratan, M. McConnell, S. Bekiranov, Implementation of a Hamming distance-like genomic quantum classifier using inner products on ibmqx2 and ibmq_16_melbourne, Quantum Mach. Intell. 2 (1) (2020) http: //dx.doi.org/10.1007/s42484-020-00017-7.
- [137] A.P. Dash, S.K. Sahu, S. Kar, B.K. Behera, P.K. Panigrahi, Explicit demonstration of initial state construction in artificial neural networks using NetKet and IBM Q experience platform, Quantum Inf. Process. 19 (1) (2020) http://dx.doi.org/ 10.1007/s11128-019-2514-0.
- [138] A. Gokhale, M.B. Pande, D. Pramod, Implementation of a quantum transfer learning approach to image splicing detection, Int. J. Quantum Inf. 18 (5) (2020) http://dx.doi.org/10.1142/S0219749920500240.
- [139] J. Romero, A. Aspuru-Guzik, Variational quantum generators: Generative adversarial quantum machine learning for continuous distributions, Adv. Quantum Technol. 4 (1) (2021) 2000003, http://dx.doi.org/10.1002/qute.202000003, arXiv:1901.00848.
- [140] B. Lee, M. Perkowski, Quantum machine learning based on minimizing Kronecker-Reed-Muller forms and Grover search algorithm with hybrid oracles, in: Proceedings - 19th Euromicro Conference on Digital System Design, DSD 2016, Institute of Electrical and Electronics Engineers Inc., 2016, pp. 413–422, http://dx.doi.org/10.1109/DSD.2016.30.
- [141] Q.C. Nguyen, L.B. Ho, L.N. Tran, H.Q. Nguyen, Qsun: an open-source platform towards practical quantum machine learning applications, Mach. Learn.: Sci. Technol. 3 (1) (2022) 015034, http://dx.doi.org/10.1088/2632-2153/ac5997.